



Guidance on social protection for people affected by tuberculosis





Guidance on social protection for people affected by tuberculosis

Guidance on social protection for people affected by tuberculosis

ISBN (WHO) 978-92-4-008932-7 (electronic version) ISBN (WHO) 978-92-4-008933-4 (print version)

ISBN (ILO) 9789220404478 (print) ISBN (ILO) 9789220404485 (web PDF)

© World Health Organization and International Labour Organization, 2024

Some rights reserved. This work is available under the Creative Commons Attribution-NonCommercial-ShareAlike 3.0 IGO licence (CC BY-NC-SA 3.0 IGO; https://creativecommons.org/licenses/by-nc-sa/3.0/igo).

Under the terms of this licence, you may copy, redistribute and adapt the work for non-commercial purposes, provided the work is appropriately cited, as indicated below. In any use of this work, there should be no suggestion that the World Health Organization (WHO) or the International Labour Organization (ILO) endorse any specific organization, products or services. The unauthorized use of the WHO or ILO names or logos is not permitted. If you adapt the work, then you must license your work under the same or equivalent Creative Commons licence. If you create a translation of this work, you should add the following disclaimer along with the suggested citation: "This translation was not created by the World Health Organization (WHO) or the International Labour Organization (ILO). Neither WHO nor ILO are responsible for the content or accuracy of this translation. The original English edition shall be the binding and authentic edition".

Any mediation relating to disputes arising under the licence shall be conducted in accordance with the mediation rules of the World Intellectual Property Organization (http://www.wipo.int/amc/en/mediation/rules).

Suggested citation. Guidance on social protection for people affected by tuberculosis. Geneva: World Health Organization and the International Labour Organization, 2024. Licence: CC BY-NC-SA 3.0 IGO.

Cataloguing-in-Publication (CIP) data. CIP data are available at https://iris.who.int/.

Sales, rights and licensing. To purchase WHO publications, see https://www.who.int/publications/book-orders. To submit requests for commercial use and queries on rights and licensing, see http://www.who.int/copyright. Information on ILO publications and digital products can be found at: www.ilo.org/publis.

WHO Photographs. WHO photographs are copyrighted and are not to be reproduced in any medium without obtaining prior written permission. Requests for permission to reproduce WHO photographs should be addressed to: http://www.who.int/copyright

Third-party materials. If you wish to reuse material from this work that is attributed to a third party, such as tables, figures or images, it is your responsibility to determine whether permission is needed for that reuse and to obtain permission from the copyright holder. The risk of claims resulting from infringement of any third-party-owned component in the work rests solely with the user.

General disclaimers. The designations employed and the presentation of the material in this publication do not imply the expression of any opinion whatsoever on the part of WHO or ILO concerning the legal status of any country, territory, city or area or of its authorities, or concerning the delimitation of its frontiers or boundaries. Dotted and dashed lines on maps represent approximate border lines for which there may not yet be full agreement.

The mention of specific companies or of certain manufacturers' products does not imply that they are endorsed or recommended by WHO or ILO in preference to others of a similar nature that are not mentioned. Errors and omissions excepted, the names of proprietary products are distinguished by initial capital letters.

All reasonable precautions have been taken by WHO and ILO to verify the information contained in this publication. However, the published material is being distributed without warranty of any kind, either expressed or implied. The responsibility for the interpretation and use of the material lies with the reader. In no event shall WHO or ILO be liable for damages arising from its use.

Contents

Foreword												. V
Acknowledgments												. vi
Abbreviations and acronyms												vii
Executive summary												VIII

1.		cial protection for people affected by TB: ncepts, definitions, and impact1
	1.1	The social determinants of TB and social protection in the context of the End TB Strategy 2
	1.2	What is social protection?
	1.3	Strategies for the delivery of social protections programmes for people affected by TB 6
	1.4	The role of social protection in ending the TB epidemic: a conceptual framework
	1.5	The impact of social protection on TB: the evidence

2. Planning social protection programmes for people affected by TB _____

2.1	Overview of the key steps for planning social protection programmes		 	 .14
2.2	Establishing a multidisciplinary/multisectoral team		 	 .16
2.3	Mapping the social protection landscape		 	 .17
2.4	Assessing the social protection needs of people affected by TB		 	 .18
2.5	Identifying barriers to social protection faced by people affected by TB		 	 .20
2.6	Synthesis of the evidence and formulation of social protection intervention	ons		 .21

3	Im	plementing TB-sensitive social protection	
		bgrammes	_23
	3.1	Promoting TB-sensitive social protection programmes	24
	3.2	Overcoming programmatic barriers to expanding coverage of social protection programmes	25
	3.3	Protecting the income security of people affected by TB	27
	3.4	Monitoring the coverage of TB-sensitive social protection programmes among people affected by TB	31

__13

Guidance on social protection for people affected by tuberculosis

4	Im	plementing TB-specific social protection	
			33
	4.1	Socioeconomic support	.34
	4.2	Nutrition support	.36
	4.3	Interventions to reduce TB-associated stigma	.38
	4.4	Monitoring the performance of TB-specific social protection programmes	.40
5.		tablishing and sustaining multisectoral laborations between TB and	
		cial protection4	41
	5.1	Multisectoral collaboration for action on the social determinants of TB: challenges and opportunities	.42
	5.2.	The multisectoral accountability framework to accelerate progress to end TB	.43
	5.3	Establishing effective multisectoral collaboration: the Health in All Policies (HiAP) approach	.44
Refere	nces		51
Annex		Immary of health-related social protection policies, idelines and guidance developed by the UN	59
Annex	2. Gl	ossary	63

Foreword

The World Health Organization (WHO) has set ambitious targets to end the tuberculosis (TB) epidemic, which include a 90% reduction in TB deaths, an 80% reduction in TB incidence, and the elimination of the catastrophic total costs incurred by people with TB and their households by 2030 compared with 2015, all in line with the United Nations (UN) Sustainable Development Goals (SDGs).

The inequities fuelling the TB epidemic have been accentuated by the COVID-19 pandemic, geopolitical instability, and global financial crises, making it even more urgent to implement policies that address the social determinants of TB. Social protection, a component of the WHO *End TB Strategy (1)*, has been upheld by WHO Member States as an essential component of the response to TB in the 2017 Moscow Declaration to End TB *(2)* and in the *2023 Political Declaration of the High-level Meeting of the United Nations General Assembly on the Fight Against Tuberculosis (3)*. This recognition is reflected in the new target in the 2023 political declaration, to ensure that 100 per cent of people with TB have access to a health and social benefits package so they do not have to endure financial hardship because of their illness. Similarly, the political declaration of the 2023 United Nations High-level Meeting on Universal Health Coverage includes a commitment to reverse the trend of rising catastrophic out-of-pocket health expenditure by providing measures to ensure financial risk protection and eliminate impoverishment due to health-related expenses, with special emphasis on the poor as well as those who are vulnerable or in vulnerable situations.

In 2021, delegates of the 109th International Labour Conference asked the International Labour Organization (ILO) to take urgent action towards universal social protection to accelerate the reduction of inequalities and address the financial and societal consequences of the COVID-19 pandemic. This call built on International Social Security Standards and in particular the *Social Security (Minimum Standards) Convention, 1952 (No. 102)* and the *Social Protection Floors Recommendation, 2012 (No. 202)*, which include the establishment of national social protection systems accessible to all to guarantee that people have effective access to health care without hardship, and to income security across the life course (4,5).

Social protection is a human right. However, most of the global population is not yet covered by any form of social protection, leaving billions of people unprotected. Likewise, most WHO Member States are not yet including social protection in the programmatic response to TB in a comprehensive and sustainable way.

Recognizing these significant challenges, the WHO Global Tuberculosis Programme and the ILO have jointly developed this technical guidance to assist countries with the adaptation and implementation of social protection programmes to maximize their coverage and impact on people affected by TB.

This guidance is intended to be implemented as part of the multistakeholder and multisectoral response to TB. The translation of social protection policies that serve the needs of people affected by TB requires a coordinated effort within and beyond the health sector, with strong accountability of all actors involved. Through this guidance providers of TB services, including staff and managers in TB programmes and in ministries of health, and relevant social protection stakeholders, will be equipped with both the conceptual and operational tools to plan and implement social protection programmes to accelerate progress towards ending the TB epidemic, while protecting the human rights of people affected by TB.

Acknowledgments

The production and technical editing of the *Guidance on social protection for people affected by TB* was led by Ernesto Jaramillo, with support from Delia Boccia and Debora Pedrazzoli, under the overall guidance of Farai Mavhunga and Tereza Kasaeva, Director of the WHO Global Tuberculosis Programme; and Lou Tessier, with support from Mathilde Mailfert, under the overall guidance of Shahra Razavi, Director of ILO's Universal Social Protection Department.

WHO and ILO acknowledge the following persons who led the drafting of specific themes within the guidance:

- Ines Garcia Baena (WHO) and Nobuyuki Nishikiori (WHO) on patient TB cost surveys;
- Annemieke Brands (WHO), Sabine Verkuijl (WHO) and Kerri Viney (WHO) on the socioeconomic impact of TB on children and adolescents;
- Delia Boccia (WHO consultant) on the planning of social protection programmes for people affected by TB;
- Mathilde Mailfert (ILO), Lou Tessier (ILO), Diddie Schaaf (ILO) and Afsar Syed Mohammed (ILO) on the planning and implementation of TB-sensitive social protection programmes;
- Daniel Carter (London School of Hygiene and Tropical Medicine) on the implementation of TB-specific social protection programmes based on material support;
- The KNCV Stigma Group (including Agnes Gebhard, Kathy Fiekert, Leva Leimane, Svetlana Pak, Emmy van de Grinten and Ineke Huitema) on TB-related stigma; and
- Nicole Valentine (WHO), Orielle Solar (Pan American Health Organization (PAHO), Carmel Williams (WHO Collaborating Centre for Advancing Health in All Policies Implementation), Michaela Told (HumanImpact5) and Claudia Galicki (WHO) on the establishment of multisectoral partnerships.

WHO appreciates the contributions and inputs of the following reviewers of the draft guidance: Marzia Calvi (WHO), Maria De Las Nieves Garcia Casal (WHO), Sarah Anderson (Public Health England, the United Kingdom), Anurag Bhargava (Yenepoya University, India), Jaime Bayona (World Bank), Yulia Chorna (Independent consultant), Brian Citro (Independent consultant), Stéphanie Dagron, University of Geneva, Anne-Celine Garfin (Department of Health, the Philippines), Knut Lönnroth (Karolinska Institutet), Mauro Sanchez (University of Brasilia), Viorel Soltan (Stop TB Partnership), Guy Stallworthy (Bill & Melinda Gates Foundation).

Financial support for the preparation and production of this publication was provided by the Government of Japan and the United States Agency for International Development (USAID).

WHO appreciates the contributions and inputs of the participants of the Online Expert Consultation on the WHO's global guidance to promote social protection into TB services 14-16 December 2021: Menonli Adjobimey (NTP, Benin), Yaw Adusi-Poku (NTP, Ghana), Mohammad Kamin Ayoubi (NTP, Afghanistan), Kenza Bennani (WHO), Vineet Bhatia (WHO), Jaime Bayona (World Bank), Sara Bernardini (World Food Programme), Anurag Bhargava (University of Yenepoya), Stela Bivol (WHO), Delia Boccia (London School of Hygiene & Tropical Medicine), Saskia den Boon (WHO), Daniel Carter (London School of Hygiene & Tropical Medicine), Yuliya Chorna (WHO Civil Society Task Force), Khun Kim Eam (NTP, Cambodia), Carlton Evans (Imperial College), Dennis Falzon (WHO), Ines Garcia-Baena (WHO), Agnes Gebhard (KNCV tuberculosis Foundation), Nebiat Gebreselassie (WHO), Sayohat Hasanova (WHO), Nguyễn Bình Hòa (NTP, Vietnam), Avinash Kanchar (WHO), Seiya Kato (Research Institute of Tuberculosis/ Japan Anti-TB Association), Corinne Merle (WHO/TDR), Carlos Van der Laat (International Office of Migration), Knut Lönnroth (Karolinska Institute), Yeraliyeva Lyazzat (National Research Center of Phthisiopulmonology, Kazakhstan), Allison Oman Lawi (World Food Programme), James Malar (Stop TB Partnership), Partha Mandal (WHO), Fukushi Morishita (WHO), Andre Ndongosieme (WHO), Nobu Nishikiori (WHO), Winston Palasi (Department of Health, Philippines), Tia Palermo (University at Buffalo), Gita Parwati (WHO), Kalpeshsinh Rahevar (WHO), Mauro Sanchez (University of Brasilia), Jamhoih Tonsing (Global Fund), Deng Serongkea (WHO), Priya Shete (University of California San Francisco), Tieng Sivanna (NTP, Cambodia), Michael Smith (World Food Programme), Guy Stallworthy (Bill & Melinda Gates Foundation), Viorel Soltan (Stop TB Partnership), Tatyana Toichkina (Ministry of Health, Russian Federation), Nicole Valentine (WHO), Kerri Viney (WHO), Tom Wingfield (Liverpool School of Tropical Medicine), Zhao Yanlin (Center of Diseases Control, China), Askar Yedilbayev (WHO), Saltanat Yegeubayeva (WHO), Adugna K. Yimam (World Vision), Hout Chan Yuda (NTP, Cambodia), Chamnan Yungthaisong (NTP, Thailand).

Abbreviations and acronyms

CBOs	community-based organizations
CI	confidence interval
DR-TB	drug-resistant tuberculosis
Hiap	Health in All Policies
ILO	International Labour Organization
MAF-TB	multisectoral accountability framework to end tuberculosis
MDR-TB	multidrug-resistant tuberculosis
NGO	non-governmental organization
NTP	national tuberculosis programme
SDGs	Sustainable Development Goals
ТВ	tuberculosis
UN	United Nations
UNDG	United Nations Development Group
WFP	World Food Programme
WHO	World Health Organization

Executive summary

Universal social protection is at the core of the 2030 Agenda for Sustainable Development and is also reflected in Pillar 2 of the WHO End TB Strategy (1), which acknowledges social protection and poverty alleviation as integral and essential parts of the global response to TB. During the *2023 High level Meeting of the United Nations General Assembly on the Fight Against Tuberculosis* (3) world leaders approved a new political declaration that re-emphasises the critical role of social determinants in perpetuating the cycle of TB and poverty and how these determinants and the socioeconomic consequences of TB are triggered or exacerbated by health and humanitarian crises, including pandemics, disasters and climate change. The 2023 political declaration further emphasises the continued importance of multisectoral action and accountability across the health, nutrition, finance, labour, education and social protection.

The need of mobilising sufficient and sustainable financing towards universal health coverage and social protection strategies in order to alleviate the health and non-health related financial burden experienced by households affected by TB was acknowledged in both the 2023 United Nations General Assembly High Level Meeting on TB and Universal Health Coverage. In particular, the one on Tuberculosis approved a new ambitious target to have, by 2027, all people affected by TB provided with health and social benefits packages to mitigate the financial hardships incurred due to the disease *(3)*.

The COVID-19 pandemic clearly illustrated the extent to which social protection systems can be stretched by global crises, and how these systems can also be leveraged to support ambitious global public health efforts. The role of social protection in this multisectoral response to TB is supported by an increasing body of evidence consistently showing the positive impact of social protection on TB incidence, treatment and catastrophic total costs (defined as direct medical expenditures, direct nonmedical expenditures and indirect costs, e.g. income losses, that sum to > 20% of household income), and, ultimately, the life and wellbeing of people affected by TB.

This guidance aims to translate existing best practices and current WHO and ILO policies into programmatic actions by providing staff and managers in TB programmes and in ministries of health, along with relevant social protection stakeholders, with:

- 1. essential notions and the base of evidence *to understand* how social protection can contribute to ending the TB epidemic;
- practical advice to plan for inclusive and locally appropriate social protection programmes able to respond to the needs of people affected by TB; and
- **3.** key steps **to implement** effective social protection strategies and establish strong and sustainable synergies across sectors.



More specifically:

- Section 1 presents social protection in the context of the *End TB Strategy (1)* and provides key operational definitions and concepts. The section presents a conceptual framework describing the way social protection is expected to impact TB, and a synthesis of the best available evidence on the impact of social protection on TB incidence, TB treatment outcomes and costs incurred by people with TB.
- Section 2 provides guidance on the planning of social protection for people affected by TB, including the mapping of the existing social protection landscape and an assessment of the needs for social protection programmes and barriers to access faced by people affected by TB. The section presents two complementary approaches to delivering social protection in the context of the TB epidemic (that is TB-sensitive and TB-specific).

Sections 3, 4 and 5 provide guidance on the implementation of social protection:

- Section 3 presents ways to enhance the impact of existing social protection programmes on TB by making them more responsive to the needs of people affected by TB. The section illustrates the role TB programmes can play in shaping the TB-sensitivity of social protection programmes by actively engaging with relevant ministries and other social protection stakeholders to increase access to these programmes by people affected by TB.
- » Section 4 provides guidance for the design and implementation of TB-specific social protection programmes that aim to address income and food security needs of people affected by TB. The section outlines generic protocols and actions to implement material and nutritional support alongside established or new monitoring and evaluation practices. The section also presents a collection of current best practices to address TB stigma, a major cause of social exclusion of people affected by TB.
- Section 5 provides guidance on the formulation and establishment of partnership models to foster collaboration between TB programmes and social protection services to address the social determinants of TB. Starting with the central role of the WHO *Multisectoral Accountability Framework to Accelerate Progress to End Tuberculosis by 2030* (MAF-TB) (8) and building further upon the Health in All Policies (HiAP) principles, the section illustrates the operational features of effective and efficient multisectoral collaborations.



Social protection for people affected by TB: concepts, definitions, and impact



Credit: ILO / Crozet Marcel

This section introduces key concepts and definitions related to the social determinants of TB and social protection in the context of the *End TB Strategy (1)*. The section also presents a conceptual framework to describe the expected impact of social protection on the TB epidemic, and possible entry points for intervention. The section concludes by discussing the extent to which the available evidence supports this conceptual framework.

1.1 The social determinants of TB and social protection in the context of the End TB Strategy

The TB epidemic is strongly influenced by social and economic development (9,10), including health-related risk factors (such as undernutrition, diabetes, HIV, alcohol use disorders, and smoking) (11), as well health systems weaknesses and inadequate investment in social protection (12).

Evidence from Europe and North America suggests that TB mortality was in widespread decline from the middle decades of the nineteenth century onwards, mainly because of wider socioeconomic and nutrition advances rather than specific medical interventions (13). The role of social determinants in the TB epidemic still stands: nutrition remains a major driver of the TB epidemic, with an estimated 2.2 million incident cases of TB attributable to undernutrition in 2021 (11). Furthermore, a 2023 analysis confirms that low human development and social protection spending are still major contributing factors to TB incidence in low- and lower middle-income countries (10).

Ending the TB epidemic requires implementing a mix of biomedical, public health and socioeconomic interventions along with research and innovation as outlined in the WHO *End TB Strategy (1,7)*. Pillar 2 of the strategy calls for bold policies and supportive systems, including social protection and poverty alleviation, to address the social determinants of TB. This vision is consistent with other global strategies that, over the past few decades, have gradually been adapted in response to the need to complement biomedical approaches with social protection programmes (see Annex 1 for a compendium of key policy guidance and recommendations developed by UN agencies in the context of social protection and health including occupational health, maternal/child health, and migrant and refugee health).

The *End TB Strategy (1)* is founded on promoting, protecting and fulfilling all human rights and the dignity of all people affected by TB. A person affected by TB is defined as "any person with TB infection or disease or who previously had TB disease, as well as their caregivers and immediate family members, and members of TB key and vulnerable populations, such as children, healthcare workers, indigenous peoples, people living with HIV, people who use drugs, prisoners, miners, mobile and migrant populations, women, and the urban and rural poor" *(14)*. One of the core principles of the strategy is the protection and promotion of human rights, ethics and equity. Aligned with the SDGs, the WHO *End TB Strategy (1)* encompasses strategic actions within and beyond the health sector and considers social protection as a human right. The political declaration of the *2023 UN General Assembly High-level Meeting on the Fight Against TB* supports the WHO Multisectoral Accountability Framework for tuberculosis, asking to establish or strengthen high-level multisectoral accountability and review mechanisms *(3)*. This framework provides a mechanism to address the broader determinants of TB through a holistic approach with multisectoral action and accountability.

The WHO *Multisectoral Accountability Framework to Accelerate Progress to End Tuberculosis by 2030* (MAF-TB) (8) is aligned with the 2030 Agenda for Sustainable Development (6), which seeks to build synergies between stakeholders of different SDGs. Fig. 1.1 illustrates how at least 12 SDGs are linked with *End TB Strategy* (1) targets and shows the linkages between social protection and the other SDG targets. In particular, the linkage between social protection and the epidemics of AIDS, TB, malaria and neglected tropical diseases and combat hepatitis, water-borne diseases and other communicable diseases) reaffirms the need for social protection for people affected by TB.

1. Social protection for people affected by TB: concepts, definitions, and impact





Fig. 1.1 Social protection (A) and TB (B) in the 2030 Agenda for Sustainable Development: relevant goals and targets

* Non-communicable diseases

Sources: (A) Building the future of social protection for a human-centred world of work. International Labour Conference 109th Session, 2021. Geneva: International Labour Organization, 2021 (https://www.ilo.org/ilc/ILCSessions/109/reports/reports-to-the-conference/WCMS_780953/lang--en/index.htm). (B) Essential Tuberculosis, Raviglione MC & Migliori GB (co-editors), chapter 38, page 354, 2021 (under authorization from the publisher Springer).

1.2 What is social protection?

Social protection refers to the set of policies and corresponding programmes designed to prevent and reduce poverty, vulnerability and social exclusion throughout the life course *(15)*. The right to social security (used interchangeably with "social protection") is reflected in Article 22 of the Universal Declaration of Human Rights of 1948 which states that "everyone, as a member of society, has the right to social security" *(16)*, and also in other human rights instruments, in particular the 1966 International Covenant on Economic, Social and Cultural Rights *(17)*. International social security standards – developed under the auspices of the ILO – complement this normative commitment by setting out principles and minimum requirements for national social protection systems. According to the Office of the High Commissioner for Human Rights, the right to social security encompasses "the right to access and maintain benefits without discrimination, in order to secure protection from:

- a lack of work-related income caused by sickness, disability, maternity, employment injury, unemployment, old age, or death of a family member;
- unaffordable health care; and
- insufficient family support, particularly for children and adult dependents" (18).

niversal social protection refers to comprehensive, sustainable and adequate protection for all, over the life cycle, along three core dimensions::

- Universal coverage in terms of persons protected means all should have effective access to social protection throughout the life cycle when needed.
- Comprehensive protection refers to the social risks and contingencies that are covered including access to health care and income security. The Social Security (Minimum Standards) Convention, 1952 (No. 102) defines the nine contingencies that all human beings may face over their life course: the need for medical care and the need for benefits in the event of sickness, unemployment, old age, employment injury, family responsibilities, maternity, invalidity and survivorship (acknowledging the needs of widows and orphans). It also includes protection against new and emerging risks, such as long-term care needs (4).
- Adequate protection means benefits provided need to be set at a level that effectively prevents poverty, vulnerability and social exclusion, maintains decent standards of living, and allows people to lead healthy and dignified lives (19).

With these broad objectives, social protection policies encompass a wide range of programmes of different types, in which benefits are social transfers financed via social contributions, taxes or a mix of those financing sources (20). Social transfers (all social protection benefits) comprise transfers either in cash or in kind; that is, they represent a transfer of income, goods or services (for example, health care or social care services). This transfer may be from the economically active to the retired, the healthy to the sick, or the affluent to the poor, among others. The recipients of such transfers may be eligible to receive them because they have contributed to such a programme (contributory programme), or because they are residents (universal programmes for all residents), because they fulfil specific age criteria (categorical programmes), meet specific resource conditions (means-tested programmes), or because they fulfil several of these conditions (21). Most social protection programmes are mandatory (for example health insurance is mandatory for all residents in many countries), but some are voluntary (21).

An overview of the different types of social protection programmes and their main characteristics is provided below, drawing on the ILO *World Social Protection Report 2020–2022 (15)*. A glossary of key social protection terms as well as other terms relevant for this guidance is provided in Annex 2. Such definitions are not universal, but they are meant to provide conceptual clarity in the context of this guidance.



Overall, social protection measures include the following types of programmes (15).

- **Contributory programmes**: programmes in which contributions made by protected persons (actual or potential beneficiaries) directly determine entitlement to benefits (acquired rights). The most common example of contributory social security programmes are statutory social insurance programmes, usually covering workers in (formal) wage employment and, in some countries, the self-employed, and in many cases also their families and categories of the population who are exempt from contributions for certain reasons (for example, due to caring for children).
 - » Social insurance: a contributory social protection programme that guarantees protection through an insurance mechanism, based on: (1) the payment of contributions before the occurrence of the contingency; (2) the sharing or "pooling" of risk; and (3) the notion of a guarantee. The contributions paid by (or for) insured people are pooled together, and the resulting fund is used to cover the expenses incurred exclusively by those individuals affected by the occurrence of the contingency. In contrast to commercial insurance, risk-pooling in social insurance is based on the principle of solidarity, with contributions typically related to people's capacity to pay (often proportional to earnings) as opposed to premiums that reflect individual risks. For example, social health insurance is typically a health insurance programme with public stewardship and at least some insurance contributions from and/or on behalf of the protected persons or some categories of protected persons.
- Non-contributory programmes: programmes that require no direct contribution from beneficiaries or their employers as a condition of entitlement to receive relevant benefits. The term covers a broad range of programmes, for example:
 - » universal non-contributory services for all residents (such as national health services);
 - » **categorical** programmes for certain broad groups of the population (such as universal pensions for all older people above a certain age, or universal child benefits for all children below a certain age);
 - means-tested programmes in which only persons below a certain income level are eligible to receive the benefit (such as family benefits targeted only at poor families);
 - » benefit-tested programmes in which only persons who are not already receiving benefits from another programme are eligible (for example social pensions for older persons not already receiving a pension from a social insurance programme because they were not able to meet the minimum contributory requirements over their working life); and
 - » **social assistance** programmes that provide benefits to vulnerable groups of the population, especially households living in poverty. Most social assistance programmes are means-tested. Some are targeted based on categories of vulnerability, while others are targeted to low-income households. They can take several forms, for example non-contributory social pensions, fee waivers and exemptions for health care, schooling or utilities. Conditional cash transfers, for example, are programmes that provide cash to families if they fulfil specific behavioural requirements. Beneficiaries may be required to ensure that their children attend school regularly, or to use basic preventive nutrition and healthcare services. Almost all countries at all income levels have at least one form of social assistance programme.

In addition, labour market or livelihoods support programmes such as public employment programmes provide employment opportunities for people who are able to work, with the aim of facilitating their inclusion in the labour market and ensuring a basic level of earnings *(22)*. For instance, public employment guarantee programmes provide a guaranteed number of days' work per year to poor households, generally providing wages at a relatively low level (typically at the minimum wage level if this is adequately defined).

In-kind benefits can include **social care programmes** for those facing social risks such as violence, abuse, exploitation, discrimination and social exclusion, and may range from psychosocial support to linkages with needed medical and social services.

1.3 Strategies for the delivery of social protections programmes for people affected by TB

In the context of TB, two social protection delivery models can be recognized: TB-sensitive programmes and TB-specific programmes *(23)*. This guidance uses the following working definitions for these two approaches, as illustrated in Table 1.1.

TB-specific social protection programmes are programmes exclusively addressed to people with TB and their households. Typically, these programmes are initiated by and are under the responsibility of TB programmes and are intended to respond to specific needs not covered by the mainstream social protection system. Such programmes have the explicit objective of reducing TB incidence, mortality and patient costs by mitigating the impact of poverty, vulnerability and social exclusion on households and individuals affected by TB. For instance, in some low- and middle-income countries, national social protection systems are at an early stage of development or may be financially constrained, and therefore they may be unable to offer benefits suitable for people affected by TB (such as cash benefits that could cover transport or nutrition costs). In these instances, TB-specific social protection programmes can help cover such needs. The budget for this type of programme typically lies with the TB programme and therefore it is important to consider progressively linking TB-specific social protection programmes to the national social protection system, with a view to ensuring their long-term financial sustainability.

TB-sensitive social protection programmes are programmes not targeted specifically at households or individuals affected by TB. While not designed intentionally to address any TB-related objective, these programmes can potentially affect the TB epidemic by including among their eligibility criteria people with, at risk of, and/or affected by the consequences of TB. Typically, these programmes are *not* initiated by TB programmes and are *not* under the direct control and management of the health sector. However, TB programmes can play a critical role in ensuring that these programmes also address the needs of people affected by TB and ultimately have a stronger impact on the TB epidemic.

	Definition	Examples
TB-specific social protection	Programmes focused on people with TB and their households with the specific aim of improving TB prevention, care and support	A monthly voucher to support TB treatment in South Africa <i>(24)</i>
TB-sensitive social protection	Programmes that are not limited to households and individuals affected by TB but that can potentially affect the TB epidemic by covering people with TB, at risk of TB, and/or affected by the consequences of TB	A conditional cash transfer programme in Brazil <i>(25)</i>

Table 1.1 Models for the delivery of social protection for people affected by TB

These two approaches are complementary for the delivery of social protection programmes and should not be considered as mutually exclusive. Depending on resources available, the local TB epidemic profile, as well as the available social protection programmes, these approaches could be considered either in combination or in a sequential manner.

Both approaches have strengths and limitations. For example, TB-sensitive social protection programmes contribute to reinforcing the social protection system as a whole and therefore address social determinants of health broadly. Universal social protection should be seen as a priority and, therefore, programmes should seek broad coverage and avoid narrow eligibility criteria. TB-sensitive social protection programmes are also usually more cost-effective because they leverage resources already in place rather than requiring a new budget to sustain administrative and delivery structures specifically for people affected by TB. Reinforcing national social protection systems can be a more sustainable approach to TB prevention, considering most TB-specific social protection programmes would target people who already have TB. Finally, they help to minimize the stigma related to TB, which can unintentionally be reinforced by TB-specific social protection programmes.

On the other hand, in most countries social protection systems – even when TB-sensitive – are often in their infancy and not yet complete in terms of both the range of benefits offered and the extent of the population they effectively cover in high TB-burden countries. Indeed, while the development and roll out of social protection policies, systems and programmes is improving in many countries, comprehensive social protection coverage is still low, as Figs. 1.2 and 1.3 illustrate (*15*).



Fig. 1.2 Coverage of effective social protection through social assistance programmes: global and regional estimates by function, 2020 (%)

 Vulnerable covered by social assistance
 28.9%

 0
 10
 20
 30
 40
 50
 60
 70
 80

 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 %
 <

Older persons

Sources: World social protection database [online database]. Geneva: ILO; 2021 (https://www.social-protection.org/gimi/WSPDB. action?id=32); based on the ILO Social Security Inquiry; ILOSTAT; national sources.

77.5%

Fig. 1.3 Percentage of vulnerable persons receiving cash benefits by country income level, 2020 or latest available year (SDG indicator 1.3.1)



Sources: World social protection database [online database]. Geneva: ILO; 2021 (https://www.social-protection.org/gimi/WSPDB. action?id=32); based on the ILO Social Security Inquiry; ILOSTAT; national sources.

Behind this global coverage average, the ILO *World Social Protection Report 2020-2022 (15)* highlights significant variations across and within regions, with Asia and the Pacific, the Arab States and Africa having pronounced coverage gaps relative to the global average. Those regions also include the majority of countries with a high burden of TB. Several high TB burden countries have tried to expand coverage of existing social protection programmes to people with TB (26).

Given the above, in many high TB burden countries, TB-specific social protection programmes may be necessary to complement existing social protection benefits and provide people with TB with social protection through TB programmes or through ad hoc social assistance programmes developed by national social protection authorities. For example, in Ecuador a conditional cash transfer was made available to people with drug-resistant TB (DR-TB), which provided cash benefits linked to adherence to treatment for up to 24 months, funded through the national tuberculosis programme (NTP) *(27)*. In some circumstances, a two-pronged approach may be considered as a strong basis for advocacy, consisting of continuous advocacy for the extension of TB-sensitive social protection while implementing TB-specific programmes and generating evidence thereof.



1.4 The role of social protection in ending the TB epidemic: a conceptual framework

The impact of social protection programmes is a result of a combination of both structural and individual-level determinants which influence their implementation, effectiveness and accessibility for those most in need, as illustrated in Fig. 1.4.

In any given context where social protection systems exist, operate effectively and are equitable and accessible for people in need, social protection is expected to affect the burden of TB by:

- reducing the risk of exposure to TB and transmission of TB;
- reducing the risk of developing TB disease; and
- reducing the risks of both poor TB treatment outcomes and adverse consequences of TB.

The impact of social protection on the burden of TB is achieved via several pathways. However, in the context of this guidance only three pathways will be considered: the material, the psychological/cognitive and the biological pathways (see Fig. 1.4).

- A. The material pathway: this pathway operates through living conditions and economic circumstances (including income, housing and working conditions, and food security) and can affect the risk of exposure to TB, the likelihood of disease progression after infection and the outcomes of TB treatment. A strong body of evidence confirms that social protection, especially in the form of cash transfers, can significantly improve people's material living conditions that are known driving factors of the TB epidemic in most countries.
- **B.** The psychological/cognitive pathway: according to this pathway, the distribution of TB in the general population is not attributable just to a differential exposure to material hazards and economic circumstances, but it may depend on feelings and emotions triggered by these inequalities, including stress, anxiety, anger, shame, distrust and depression. These negative emotions can translate into an alteration of psycho-neuro-endocrine mechanisms that ultimately make people more vulnerable to developing TB infection or diseases. By affecting people's mental health and wellbeing, social protection can alter these neuro-endocrine mechanisms and ultimately influence beneficiaries' vulnerability to diseases, including TB.
- **C.** The biological pathway: while not yet fully demonstrated, it can be speculated that social protection benefits, and in general, income support interventions, may affect genetic, physiologic and immunologic processes that in turn influence the likelihood of developing TB infection and TB disease, as well as the outcomes of TB treatment.

These three pathways that influence the impact of social protection on TB burden likely happen simultaneously, and collectively can influence people's behaviours and the probability of developing comorbidities. This in turn can affect TB prevention, care and consequences and ultimately the overall burden of TB in a given community. All the above pathways operate across an individual's life course and social protection may have a differential impact across critical life stages depending on the type of benefits delivered and when the social protection benefits are provided. In particular, social protection programmes can play a role in reducing the burden of TB through the life course by ensuring a healthy start for children, promoting labour market participation and social inclusion during adulthood, and ensuring secure aging among the elderly.



Fig. 1.4 Understanding the impact of social protection on TB: a conceptual framework

1. Social protection for people affected by TB: concepts, definitions, and impact

1.5 The impact of social protection on TB: the evidence

This section provides an overview of the evidence base for the impact of social protection programmes on several *End TB Strategy (1)* targets, including TB incidence, TB treatment outcomes, and people with TB and their households experiencing catastrophic total costs. Evidence related both to the impact on TB prevention (that is, initiation and adherence to TB preventive treatment, as well as the uptake and efficacy of Bacillus Calmette-Guérin vaccination) and the impact on TB diagnosis, is very limited, therefore will not be presented in this guidance despite its critical importance.

1.5.1 The impact of social protection on TB incidence

Modelling studies confirm that increasing social protection coverage can significantly contribute to improvement in key TB indicators. A global ecological study estimates that social protection spending levels are inversely associated with TB incidence, prevalence and mortality (12). Similarly, a modelling study suggests that expanding social protection coverage to all those in need could result in a 9% annual decline in TB incidence, and a cumulative reduction in TB incidence of 76.1% by 2035 (28).

1.5.2 The impact of social protection on TB treatment outcomes

Material support, in the form of food or financial enablers is recommended by WHO for the care and support of people with DR-TB (29). A growing body of evidence also confirms the positive impact of social protection on treatment outcomes of people with drug-susceptible TB. Three systematic reviews conducted during the period 2018–2021 confirm the positive effect of social protection, mainly in the form of social assistance, on TB treatment outcomes in a variety of low- and middle-income settings (30-32). This effect is consistent across studies, although its magnitude may be context specific.

Financial incentives as low as \$US 1 have been shown to improve TB treatment outcomes (33-38). Cash transfers have been particularly studied in Latin America, with positive impacts (up to a 10% increase in TB treatment success) shown across the region (25,39-46). Food baskets or vouchers – sometimes delivered in combination with other interventions – also demonstrated positive impacts in the similarly middle-income settings of Brazil, Russia, Peru and India (46-54). Transportation vouchers also showed positive but smaller impacts on TB treatment outcomes in two studies in China (51,52). Notably, interventions in Peru, Nepal and Thailand suggest that their impact can be enhanced when paired with counselling, psychological and other forms of material support ("cash plus" approaches) (35,44,53,54).

1.5.3 The impact of social protection on patient costs due to TB

A pooled analysis of findings from 27 national TB patient cost surveys showed that the percentage of people with TB and their households that experienced catastrophic total costs (defined as direct medical expenditures, direct nonmedical expenditures and indirect costs, e.g., income loss that sum to >20% of household income) ranged from 13% to 92% with a pooled average of 48%: that is, about half of all people affected by TB faced catastrophic total costs (*11*). These figures have been further confirmed in a recent meta-analysis (*55*). Finally, increasing evidence suggests that TB exerts a severe socioeconomic impact also on children and adolescents affected by TB, as Box 1.1 illustrates.

These figures are alarming, considering the *End TB Strategy* target (1) of zero catastrophic total costs and the known association of these catastrophic costs with poor TB treatment outcomes (56). Unlike the health expenditures considered in the related SDG indicator 3.8.2 that are restricted to direct health payments (also known as out-of-pocket heath payments and concern the whole population), costs faced by people with TB encompass *medical costs*, including in settings where TB diagnosis and treatment are provided free of charge, *direct non-medical costs* (such as transportation, accommodation, food and nutritional supplements) and *indirect costs* (such as productivity and economic costs incurred by a patient or household due to TB healthcare visits and hospitalization

during the TB episode). Although the composition of cost drivers differs by country, direct non-medical and indirect costs consistently represent the largest proportion of the economic burden of TB on individuals and households. Accordingly, simply accelerating universal health coverage policies (that is, by making TB care services available and accessible geographically and financially through, for example, the provision of free TB care services and/or the expansion of health insurance) may be insufficient to completely defray costs (*57*).

The greatest costs incurred by people with TB occur early in their interactions with the healthcare system, suggesting that social protection programmes may need to be delivered at the time of diagnosis to eliminate catastrophic total costs (*37,57,58*) and improve TB treatment outcomes. Depending on the specific type of programme considered, social protection is expected to mitigate catastrophic total costs either by defraying direct medical costs (through health insurance mechanisms) or by compensating any income loss due to inability to work and/or unexpected household expenditures (e.g. through sick leave benefits and cash transfers). Transport and food costs are commonly cited as non-medical costs incurred by people with TB, and a useful TB-specific intervention is to provide transport and food vouchers – even amounts as small as 3–6% of household income have been demonstrated to improve TB treatment outcomes (*31*).

Evidence from Peru suggests that 30% of households affected by TB receiving social protection incurred TBrelated catastrophic total costs; compared to 42% of households affected by TB who were not receiving social protection benefits (*59*). This effect also translated into improved TB treatment outcomes (*60*). Similar findings are available from Indonesia, where the chances of an unsuccessful treatment outcome were up to four times higher among patients who experienced catastrophic costs compared with those who had not experienced catastrophic total costs (*56,61*). Finally, a 2017 economic model has demonstrated that both TB-sensitive and TB-specific social protection programmes can partly defray catastrophic total costs across a variety of settings in low- and middle-income countries (*62*).

Box 1.1 The socioeconomic impact of TB on children and adolescents

A 2021 sub-analysis from 10 national TB patient cost surveys, including almost 1,500 children and adolescents, provided valuable insights on the financial burden of TB in households with this age group. This analysis showed that if the person with TB was a child or an adolescent, the proportion of households that experienced catastrophic total costs was lower (41.8%, 95% confidence interval (CI) 22.9–60.8%) than if the person was an adult (56.2%, 95% CI 44.4–68.1%) *(63)*.

School disruption was one of the major consequences noted among children (8.4%, 95% CI: 3.4–13.4%) and adolescents with TB (18.7%, 95% CI: 8.8–28.7%). Food insecurity was experienced by 19.8% of children with TB (95% CI: 3.7–35.8%) and 20.5% of adolescents with TB (95% CI: 11.5–29.8%).

Findings from a 2022 scoping review, including 36 different studies, confirm and further add to this analysis *(64)*. In this review, the socioeconomic impact exerted by TB ranged from impoverishment, stigma and family separation to effects on nutrition and missed education opportunities. None of the studies included in this review provided sufficient follow-up to observe the long-term impact of TB on children and adolescents, and how TB can affect the life and work trajectory of this group, leaving a substantial gap in the understanding of the life course implications of developing TB during childhood or adolescence.

Overall, these findings emphasize the need for providing social protection for children and adolescents affected by TB. However, there is still limited knowledge of the extent to which different social protection programmes (in-kind, in-cash, or a combination of both approaches) can impact the short- or long-term socioeconomic consequences of TB, as well as how child-sensitive social protection programmes could support households affected by TB.





Credit: WHO / Lindsay Mackenzie

The adoption of effective and efficient social protection measures for people affected by TB requires proper planning, encompassing the assessment of what social protection system and programmes are locally available, the understanding of the social protection needs of people and communities affected by TB, and finally the accurate identification of the individual- and structural-level barriers that people affected by TB face in accessing existing social protection programmes.

This section outlines the key steps of this planning phase: the proposed methodology is meant to support decisions and actions of TB programmes to design and implement locally appropriate social protection approaches. Specifically, this planning effort is key to identifying what the existing social protection system can offer, and what complement, if any, may be needed to meet the special needs of households affected by TB. Through this planning exercise, TB programmes can devise the most appropriate social protection investment (that is, TB-sensitive, TB-specific or a combination of both) that can most effectively and efficiently reduce the burden of TB in their setting.

2.1 Overview of the key steps for planning social protection programmes

The planning of effective and cost-effective social protection programmes is context specific and should be informed by a formal process comprising the following five steps:

- 1. establishing a multidisciplinary/multisectoral team
- 2. mapping the social protection landscape
- 3. assessing the social protection needs of people affected by TB
- 4. identifying barriers to social protection faced by people affected by TB
- 5. formulating solutions.

There are different approaches to conducting this planning exercise. The strategy chosen as well as the frequency of the activities can be calibrated based on resources available and the existing knowledge gaps. The methodology presented here draws largely upon the following resources:

- the HIV and social protection assessment tool developed by UNAIDS as part of the Inter-Agency Task Team on HIV-sensitive social protection for the identification of barriers to access (65); and
- the ILO global guide on social protection assessment-based national dialogue (66), the United Nations Development Group (UNDG) global toolkit on joint work on social protection (67) and the Interagency Social Protection Assessment (ISPA) Core Diagnostic Instrument (68) for the mapping and rapid assessment of existing social protection programmes.

Overall, this process should enable the understanding of existing challenges and opportunities that may affect availability and suitability of social protection programmes for people affected by TB and that may also ultimately influence the impact of these social protection programmes on TB. The knowledge and evidence generated through this assessment will inform the design and implementation of the social protection approach (that is, whether TB-sensitive or TB-specific) that best meets the needs of people affected by TB while respecting ethical, equity, feasibility and sustainability criteria. The breadth and scope of this planning exercise is dependent on the human and financial resources available, the sustainability requirements, and the extent of communication and integration across the health and social protection sectors. It is important to note that the assessment of the social protection system may already be available and/or have been conducted by social protection working groups, therefore TB programmes are encouraged to investigate as to whether that is the case and how they could build upon such processes.

Each of the proposed stages is adaptable to the local context and can be broken down into further steps. The next sections describe the process in detail, including objectives, expected outputs, required tools and useful references for each step, as illustrated in Fig. 2.1.



Fig. 2.1 Summary of key steps for the planning of social protection programmes for people affected by TB: the role of the coordinating team

Steps	Objectives	Activities	Outputs	Useful references and tools
Establishing a multi-disciplinary and multi- sectoral team of stakeholders	To identify and engage with key stakeholders from the TB and social protection sector	Stakeholder mapping and agreement on the targeted engagement strategy	A clear and agreed planning roadmap	Guidance for National Strategic Planning for Tuberculosis (69)
Apping the social protection landscape	To systematically collect and appraise information (mainly in terms of coverage and adequacy) on the available social protection programmes operating in the country (whether TB-sensitive or TB-specific)	 Desk review of existing literature Key informant interviews Secondary analysis of existing data 	A list of 2–3 priority social protection programmes to be further explored for their TB- sensitivity (their responsiveness to and inclusiveness of people affected by TB)	 ILO Assessment Based National Dialogue guide (66) UNDG global toolkit on joint work on social protection (67) ISPA Core Diagnostic Instrument for the mapping and rapid assessment of existing social protection programmes (68)
3 Identifying the social protection needs of people affected by TB	To describe the social protection needs of people with TB along the continuum of care as well as the social protection needs of all their household members that may be indirectly affected by TB (including children, adolescents, caregivers, elderly)	Data collection on social protection needs, ideally at different points in time (that is, at the start of treatment and monitored until treatment completion)	A clear understanding of the type, size, timing and frequency of the social protection benefits needed by people affected by TB	• Tuberculosis patient cost surveys: a handbook (70)
Identifying the barriers to access to social protection faced by people affected by TB	To assess the supply- and demand-side barriers hampering access to social protection programmes by people affected by TB	Triangulation and integration of data from: • Quantitative survey among key stakeholders from the multidisciplinary and multisectoral team • Focus group discussions with key multidisciplinary and multisectoral stakeholders	A clear understanding of the obstacles and bottlenecks hampering a wider and more equitable access to social protection programmes by people affected by TB	HIV and social protection assessment tool. Generating evidence for policy and action on HIV and social protection* (65)
D esigning appropriate social protection programmes	To discuss the findings from the previous steps and design locally appropriate social protection programmes for people affected by TB	Consultation with key members of the multidisciplinary and multisectoral stakeholders	Consensus on the most appropriate social protection strategy to maximize the impact on TB in the country, while acknowledging its cost, feasibility and acceptability	Guidance for National Strategic Planning for Tuberculosis (69)

* This tool could be adapted to best meet the objectives of the assessment

2.2 Establishing a multidisciplinary/multisectoral team

Under the leadership and coordination of the TB programme, a core group is needed to coordinate all stages of the planning exercise. This coordinating team will be responsible for:

- A. outlining the TB-related objectives to be prioritized and achieved through social protection;
- B. identifying the key literature and sources of information to map the local social protection landscape;
- C. coordinating the assessment of social protection needs and barriers to access to social protection services; and
- **D.** finalizing the social protection strategy that can most effectively impact TB in the country, based on the findings from the planning process.

The coordinating team should be supported and complemented by the knowledge and expertise of a multidisciplinary and multisectoral group of key stakeholders (maximum 15–20) engaged at various stages of the planning process. Stakeholders in this context are defined as people or groups who are affected by, can influence or may have an interest in a strategic activity *(69)*.

To ensure adequate representation of skills and expertise, this multidisciplinary and multisectoral group should include representation from the following domains:

- TB care staff (that is, nurses, doctors, community health workers and other care providers);
- civil society and communities affected by TB and TB comorbidities, and advocates of the social determinants of TB;
- the ministry of social affairs or social security;
- key policy and implementation partners, including UN agencies (such as ILO, World Food Programme (WFP), International Organization for Migration (IOM), United Nations International Children's Emergency Fund (UNICEF)) and non-governmental organizations (NGOs);
- the national statistics office; and
- others as necessary based on country needs.

Mapping the appropriate stakeholders is a dynamic process which can start with a predetermined list of core experts who, through their networks, may help to identify other relevant stakeholders to invite to be part of the group. The approaches to inviting stakeholders to take part to this planning exercise will vary depending on the role and seniority of the stakeholder but will also depend on whether these experts have already been engaged with TB-related activities and/or have had previous contacts with the TB programme.

The principles of meaningful engagement with the identified stakeholders build upon a number of principles that have been described elsewhere, including the importance of an early engagement with stakeholders in the process and throughout all its steps, the need for engaging with stakeholders as equals, and the role of existing collaborative platforms when establishing stable and close partnerships *(69)*.

Overall, the strength of this multisectoral collaboration will heavily depend on the capacity of the TB programme to identify in advance how the planning of social protection programmes for people affected by TB can also potentially progress the agenda of the respective stakeholders. Stakeholders are likely to be more actively involved if this engagement contributes also to the achievement of their own objectives *(69)*.



2.3 Mapping the social protection landscape

Mapping the local social protection landscape is essential to understanding the extent to which existing programmes are responsive to the needs of people affected by TB and the extent to which they are inclusive enough to reach those most in need. In settings where mapping of the social protection landscape is already planned or underway, efforts should be made to ensure that social protection for people affected by TB is streamlined in these exercises.

Mapping of the social protection landscape requires an in-depth review of the existing published and unpublished literature, including peer-reviewed articles, reports and policy briefs. People affected by TB are usually confronted with several social stressors; thus, when undertaking this social protection mapping exercise, the following programmes and benefits should be prioritized:

- cash or in-kind benefits that address key risk factors for TB, including income and food insecurity, poor housing and unemployment;
- healthcare benefits that cover the cost of medical care, and that cover both transportation and nutrition needs, as well as long-term care if needed;
- sickness cash benefits and, when the disease results in permanent loss of function to one's health, disability benefits to compensate income loss due to the temporary inability to work or the opportunity cost of seeking care;
- occupational diseases benefits (often part of employment injury insurance programmes) to compensate for the loss of income and restore the health of people who contracted TB through their work (such as health workers, for example);
- income support for caregivers and children in affected households (including survivors' benefits) to compensate for the opportunity cost or any income loss due to care-giving responsibilities within the affected households; and
- access to resources to counteract and be protected from the social exclusion that can result from the stigma attached to TB, TB comorbidities (such as HIV and mental disorders) and being a beneficiary of social protection.

The scope of this desk review should include a critical appraisal of the priority programmes and benefits along the following dimensions:

- *coverage*: the extent to which the eligibility criteria of existing social protection programmes result in inclusion of people affected by TB at national or sub-national level;
- adequacy: the extent to which the benefit packages provided meet the needs of people affected by TB, in terms of scope, type, size, frequency, place of delivery as well as delivery modality (assessing adequacy of benefits against the needs of people/households affected by TB is of particular importance because they often incur costs related to the disease that are very high and that existing social insurance or social assistance programmes may not entirely cover (26,71); and
- *implementation and operational issues:* the extent to which, for example, insufficient funding, limited geographical access of the institution running the programme, or administrative bottlenecks hamper the enrolment into the programme and/or access to the benefits.

International human rights instruments and international social security standards provide useful references and benchmarks for assessing coverage and adequacy of social protection benefits (see Box 2.1). For example, ILO Conventions on social security establish minimum levels of benefits (72).

Box 2.1 International human rights instruments and social security standards resources

- ILO Social protection data dashboards monitor coverage across countries and regions (73);
- ILO World Social Protection Report provides a global overview of recent developments in social protection systems (15);
- · ILO Social Protection Monitor announces social protection measures throughout the world (74);
- ILO Social protection response to the COVID-19 crisis and bi-monthly outlook (75,76);
- International Social Security Association (ISSA) Social Security Programs Throughout the World series highlights social security programmes in 183 countries and territories (77); and
- The UNDG Social Protection Floor Initiative aims to integrate social protection floors into national, regional and global development strategies (67).

Data and evidence gathered through the desk review could be further consolidated and complemented by:

- analysis of secondary data if available and/or
- interviews with key informants from the multisectoral and multidisciplinary stakeholders.

Overall, these additional sources of information could be used to address persisting knowledge gaps, including:

- cost-effectiveness and equity of the local social protection programmes;
- evaluation of the impact of existing social protection programmes on health outcomes and/or health seeking behaviour and/or relevant TB risk factors; and
- examples of multisectoral collaborations, including participation and engagement of these programmes with public or private health sector players.

The expected output from this mapping exercise is the identification of 2–3 priority programmes to further investigate for their suitability for, and accessibility by people affected by TB.

2.4 Assessing the social protection needs of people affected by TB

Social protection needs of people with TB vary across the continuum of care and need to be considered in their entirety. Half of costs incurred by people with TB are experienced before starting TB treatment and most of these costs are opportunity costs linked to loss of employment or income because of inability to work *(58)*. Increasing evidence shows that TB-associated disabilities may also affect people beyond the completion of TB treatment, making TB survivors unable to return to work either in the short or long term. Understanding the social protection needs along the continuum of care, including the needs of members of a household affected by TB (especially women, children and adolescents, elders, and those on palliative/end-of-life care) will be key to identifying who to best target with social protection programmes and when these should be provided.



Data on social protection needs can be collected through questionnaires administered by purposely trained staff to determine the health and socioeconomic challenges people with TB are likely to experience, including:

- individual and household socio-economic position (measured through econometric or asset-based indices) (78)
- sources of income (employment status in the formal or informal economy)
- education (years of schooling, literacy)
- food insecurity (see section 4.2)
- other relevant social determinants of TB (such as housing, living in slums, migration status)
- TB comorbidities and TB-associated disabilities (79)
- TB-associated stigma (see section 4.3), and
- receipt of any form of social protection.

The questionnaires may be administered at the start of treatment, during it and at the end to monitor how social protection needs change over the continuum of care. In addition to people with TB, the assessment of needs may also be extended to their household members that are indirectly affected by TB (such as caregivers, women, elderly people, children and adolescents). This could facilitate provision of social protection benefits with potentially high preventive impact considering the high risk of TB among recent contacts of someone with TB.

Alternatively, data on social protection needs may be inferred from information collected as part of ongoing or completed patient cost surveys (see Box 2.2).

Box 2.2 Assessing social protection needs: the role of TB patient cost surveys

TB patient cost surveys measure the direct and indirect costs incurred by people with TB and their households to inform policies to mitigate the financial hardship and economic burden imposed by TB on these households. WHO has established standard methods for conducting national surveys to assess these costs *(80)*.

TB patient cost surveys could be a useful entry point to investigate the social protection needs of people affected by TB and can serve as tools to advocate for the extension of coverage of existing social protection programmes to people affected by TB.

Data from a TB patient cost survey may be used to identify sub-populations to be prioritized and the interventions needed to mitigate costs and reduce or eliminate the proportion of households experiencing catastrophic total costs. Data from these surveys may also be used to infer information on the size and type of support needed by households affected by TB. Data such as household socio-economic status, nutritional and food security, occupation, ongoing receipt of any form of social protection benefits, and drivers of patient costs as well as coping mechanisms adopted in response to health expenditures, can inform the design and implementation of TB-specific social protection programmes. The analysis of this data can further contribute to improving the co-programming and reciprocal referral system across local TB care and social protection programmes.

2.5 Identifying barriers to social protection faced by people affected by TB

There are various supply-side and demand-side barriers to accessing social protection faced by people affected by TB.

Supply-side barriers: These are policy or programmatic barriers that hinder access by implicitly or explicitly excluding populations at high risk of TB (such as migrants, the homeless, people with substance abuse disorders, or people living with HIV) or households affected by TB, by limiting coverage, or by not providing sufficient or timely benefits. Programmatic barriers include lengthy or inefficient administrative procedures that may discourage enrolment or delay access to benefits. People at high risk of TB may also be excluded from access because of stigma or discriminatory practices: migrants, ethnic minorities, indigenous people, substance users, homeless, and other marginalized groups may face challenges accessing local social protection services even when they are eligible. Finally, sometimes eligibility criteria based only on income or other econometric measures (such as purchase power or monthly expenditures) may not be sufficiently sensitive to capture people's vulnerability and can lead to the exclusion of people in need (*40*).

Demand-side barriers: These are barriers that may hamper access to social protection because of individual level factors. They include, for example, lack of information about existing social protection programmes, lack of national identity cards, not having a permanent address, lack of documentation of residence status, as well as not having a bank account. While social protection programmes are free at the point of use, the process of accessing them may impose costs that are unaffordable to eligible recipients, such as travel costs, administrative costs, and waiting times resulting in hours of work lost.

The assessment of the above barriers requires collecting information to identify:

- 1. the population groups at high risk of TB in the country, and the social determinants driving the local TB epidemic;
- 2. whether people affected by TB may be eligible for the priority programmes identified through the desk review and to what extent (that is, their estimated coverage among people affected by TB, if available); and
- **3.** the main demand- and supply-side barriers limiting access to these programmes by people affected by TB.

These data can be collected through:

- a bespoke quantitative survey among a selected list of stakeholders such as staff from social protection and TB programmes, social workers, civil society, and communities affected by TB (while there is currently no standard tool for this assessment the HIV-sensitive social protection assessment tool developed by UNAIDS (65) may be adapted for this purpose); and
- a qualitative focus group discussion with selected key informants to discuss and share their opinions and knowledge (this may require some planning, higher costs and the involvement of researchers with expertise in qualitative methods).

Ideally both approaches should be undertaken sequentially: after the collection of quantitative data through the surveys, focus group discussion with key informants can be used to validate and further discuss the information obtained through the questionnaire. While more time consuming, this triangulation of qualitative and quantitative information may significantly enhance the understanding of the actual or perceived barriers to social protection experienced by people affected by TB.



2.6 Synthesis of the evidence and formulation of social protection interventions

Evidence resulting from the above steps should be synthetized and appraised to inform the design and implementation of locally appropriate social protection interventions to best address the needs of people affected by TB. Emerging themes and possible solutions can be shared during consultation with the multidisciplinary and multisectoral team of stakeholders, aiming to:

- comprehensively list social protection options emerging through the assessment process;
- facilitate discussion and understanding of the feasibility and pros and consof the various social protection options;
- reach consensus on interventions to improve accessibility and responsiveness of the social protection options and maximize their impact on people affected by TB; and
- devise a strategy for action, including the identification of actors to engage (for instance, agencies and ministries implementing specific social protection schemes).

This consultation will require appropriate representation of all key stakeholders, particularly representatives of affected communities and civil society. The discussions should focus on:

- identifying and prioritizing interventions considering existing policies and available resources, through an approach that is TB-sensitive, TB-specific, or a combination of the two;
- defining goals and objectives of the interventions which may imply defining a broader long-term goal and a series of more specific objectives that collectively can contribute to achieving the long-term goal (an example of a long-term goal would be to ensure that no TB-affected households face catastrophic costs, which may be achieved by ensuring that 100% of people with TB who are eligible for social protection are referred to the most appropriate social protection services and supported to enrol into the social protection system); and
- discussing key criteria that may affect the impact of the proposed interventions, including:
 - magnitude and type of the identified needs and barriers to access for example, what proportion of people with TB are in need of social protection and can they be realistically supported either through TB-sensitive or TB-specific social protection programmes;
 - 2. effectiveness of the intervention that is, the likelihood that the intervention will lead to the expected impact;
 - 3. equity of the intervention for example, whether the intervention is likely to benefit those most in need;
 - acceptability of the intervention such as the extent to which both implementers and target populations are likely to consider the intervention appropriate based on anticipated or experienced cognitive and emotional response to the intervention (81);
 - 5. feasibility of the intervention for example, the practicality and adequacy of the logistical and operational requirements to deliver the intervention, including costs of the intervention; and
 - **6.** sustainability over time of the intervention that is, the likelihood that the intervention is taken up by the social protection system and can be implemented by the responsible agencies over time.

All the steps described in this section collectively contribute to the formulation of locally appropriate and potentially effective social protection programmes for people affected by TB. From piloting work in countries in the WHO Western Pacific region, it is estimated that the completion of the proposed planning process in a given country may require an average of 2–3 months of work for the coordinating team and approximately US\$ 30 000, primarily for any consultancy fees to undertake the desk review of existing social protection programmes, and for any costs associated with the planning and organization of the focus group discussion (if conducted in person). Should

these resources not be available, countries may use a simple decision process like the one outlined in Fig. 2.2. This approach does not necessarily require the involvement of external stakeholders and/or the formal acquisition of new data through qualitative or qualitative methods. Instead, staff of the TB programme may embark on this planning exercise based on their best knowledge and understanding of the social protection floor in their own country. While less accurate, this approach may be sufficient in countries where either the social protection and TB sector are already collaborating and examples of multisectoral collaborations are well documented (that is, TB-sensitive programmes have been already tested); or where the limited social protection floor leaves little room for the expansion of the scope and remit of the existing programmes (in other words, TB-sensitive programmes are not an option).

The following sections provide guidance on the next steps once a strategy to meet the social protection needs of people affected by TB has been devised, whether TB-sensitive, TB-specific or a combination of both.





22



Implementing TBsensitive social protection programmes



This section describes how existing social protection programmes can be leveraged and adapted to make them more sensitive to the needs of people affected by TB. The section builds upon the example of income security and applicable labour protection norms. In particular it illustrates how TB programmes can engage with the social protection sector and overcome important underlying constraints – such as informal employment or discriminatory labour legislation and employment policies – which may significantly limit access to existing social protection benefits by people affected by TB.

3.1 Promoting TB-sensitive social protection programmes

TB-sensitive social protection programmes can be utilized to provide social protection for people affected by TB. Planning and implementation of TB-sensitive social protection programmes requires close collaboration between TB programmes and the social protection sector (see Box 3.1).

Box 3.1 Promoting the TB-sensitivity of existing social protection programmes in TB services TB programmes can promote the TB-sensitivity of existing social protection programmes by:

Advocacy

A.

toral response to TB with accountability of all key representatives from the social protection sector and strong leadership from civil society representatives – for example, TB programmes could advocate for expansion of the right to social protection for people affected by TB.

Advocating for a multisec-

Β.

Raising awareness about the role of social determinants of TB in driving the TB epidemic, and positioning social protection as a human right of people affected by TB – for example, TB programmes may help people with TB to be aware of the importance of social determinants in the epidemiology and natural history of TB as well the social protection

C.

Engagement

Engaging with the social protection sector (such as ministries of social affairs or social security) and other relevant stakeholders within and beyond the health sector to influence the design, delivery and implementation of social protection programmes – for example coverage, level of benefit provided or eligibility criteria.

D.

Securing funding to complement existing social protection benefits to address special needs of TB-affected people. ntegration

Ε.

Strengthening the integration of TB and social protection programmes to optimize the implementation of bilateral screening initiatives and to identify effective referral mechanisms across TB and social protection programmes – for example, staff working in TB programmes or social workers could be capacitated to assess people with TB for eligibility for social protection and support them to navigate the administrative and legal systems to access the relevant benefits.

Funding securement


3.2 Overcoming programmatic barriers to expanding coverage of social protection programmes

Programmatic barriers include administrative and operational barriers that limit access to existing social protection programmes by eligible beneficiaries. Possible interventions to address these barriers are informed by the context and underlying causes. Box 3.2 provides examples of approaches to improve the accessibility of these programmes for people affected by TB.

Box 3.2 Expanding social protection coverage for people affected by TB

TB programmes can support the coverage expansion of existing social protection programmes by:

А.

Negotiating with local stakeholders responsible for social protection policies and their implementation to:

- revise the eligibility criteria adopted by the programmes (Box 3.3 provides examples of how access to existing social protection programmes can be enhanced by including TB as one of the eligibility criteria of the programme);
- 2. expand geographical coverage of the social protection programme to areas where the burden of TB is higher; and
- 3. simplify and enable administrative and logistic processes to enrol for social protection services (this may involve collaboration with relevant offices to facilitate the process for people with TB to obtain required documents, such as national identity cards, or systematically collecting basic.

Β.

Awareness

Creating awareness among people affected by TB of the social protection services available and promoting social protection as a human right.

C.

Developing training materials for staff in charge of planning, delivery and administration of social protection programmes to improve their awareness of the social protection needs of people affected by TB.

Collaboration and partnership between social protection partners and community-based organizations (CBOs) can help to address some of the programmatic barriers. CBOs are close to the community, and they can ensure that the voice of people affected by TB in need of social protection is properly represented and acknowledged. They can support the identification of eligible people and promote awareness of social protection entitlements and procedures. Finally, CBOs can also contribute to the establishment of adequate and effective bilateral referral mechanisms across the TB and social protection sectors. Existing coordination and collaboration systems for TB programmes and civil society organizations can be leveraged to improve access to existing social protection programmes for people affected by TB, and to improve their responsiveness to the needs of people affected by TB.

Box 3.3 Examples of strategies to expand coverage of social protection programmes

Several countries have introduced initiatives to expand coverage of existing social protection programmes to people with TB, using various strategies to increase access as the following examples show.

- Granting access to existing disability benefits: disability benefits are short- or long-term cash benefits
 provided as full or partial income replacement to those unable to engage in paid work due to a disability
 (mostly permanent). In South Africa, a short-term, universal disability grant is provided to individuals of
 working age if they are deemed to be permanently or temporarily unable to work because of a physical
 disability or chronic illness such as DR-TB (82). However, people with TB still face challenges to accessing
 benefits, including limited awareness, timing of the benefit and the requirement for assessment by a
 specific doctor who may be based in a different health facility (83,84).
- Granting access to social assistance programmes to households with at least one member affected by TB: This involves establishing new eligibility criteria to receive benefits based on having a household member with a long-term or chronic disease, as was the case of the programme *Bono Joaquim* in Ecuador (85).

When exploring avenues to expand social protection services for people affected by TB, it is important to consider the following aspects.

- In most high TB burden countries, social protection systems and programmes are largely underfunded, which limits their scope for further expansion. There is therefore need for advocacy and resource mobilization for adequate, equitable and sustainable financing of social protection.
- 2. The design of social protection programmes has profound implications on their coverage for populations at high risk of TB, through key variables including:
- the target population and geographic coverage of the programme;
- the adequacy of social protection benefits (including the range of benefits covered and the extent of financial protection);
- the inclusion of workers in the informal economy and their families, who often face difficulties in accessing social protection programmes; and
- the inclusion of adequate sickness benefits that are not based on employer's liability (whereby the employer is required to continue paying the wage of his employees while the worker is not able to work).

Addressing these issues requires engagement of TB programmes with the social protection sector to adapt existing systems to increase access to social protection services by people affected by TB. These changes and adaptation may include:

- extending the eligibility in specific areas where people affected by TB are more likely to be resident;
- topping-up existing benefits to meet the needs of people with TB, usually through funding from international and local funding bodies; and
- increasing awareness of the multiple returns of social protection beyond TB, including positive effects on poverty and on TB-related comorbidities and risk factors such as HIV, diabetes, and undernourishment.



3.3 Protecting the income security of people affected by TB

As highlighted in Section 2, people affected by TB are vulnerable to impoverishment due to: (i) medical costs; (ii) non-medical costs (such as transportation to/from healthcare facilities, additional nutrition costs); and (iii) income loss. In low- and middle-income countries, income loss represents on average 60% of the total costs associated with TB care *(58)*.

Loss of employment and income is one of the major causes of economic burden in households affected by TB (see Box 3.4) and can be caused by:

- the absence and/or insufficient implementation of legal provisions for social protection benefits in case of sickness, resulting in people with TB losing their income when unable to work because of care seeking or sickness;
- TB not being considered as an occupational disease (such as for health workers, miners, etc.) and the consequent lack of appropriate benefits to cover employment injuries; and
- discriminatory practices against people with TB in the workplace, resulting in the termination of their employment.

Loss of income and employment is more likely among those in informal employment, which represents 70% of total employment in low- and middle- income countries (86). Informal employment refers to working arrangements that are de facto or by law not subject to national labour and social security legislation, income taxation or entitlement to social protection or certain employment benefits (87). It includes both wage employment and self-employment. Workers in informal employment are defined as those who work in informal jobs, whether carried out in formal sector enterprises, informal sector enterprises, or households (87). A significant proportion of the poor are in the informal economy.

People with TB who work in the informal economy are usually not included in the labour and social protection systems and typically have limited access to social insurance and other forms of benefits that secure their income when they are unable to work. Therefore, measures to extend social protection coverage to those in the informal economy are essential. Further, transitioning to the formal economy is highly desirable as it contributes to ensuring that TB is contained at the workplace and that people with TB can access social protection without discrimination *(88)*.

A range of labour policies to protect the employment and incomes of households affected by TB are applicable and potentially impactful. Possible entry points include measures to:

- 1. prevent the loss of employment through non-discrimination laws, labour code, and labour inspection;
- 2. use the workplace for TB prevention activities; and
- **3.** ensure protection in case of incapacitation or job loss through unemployment insurance, and sickness and disability benefits.

Box 3.4 Impact of TB on employment of people with TB in the Democratic Republic of Congo and the Lao People's Democratic Republic

The first nationally representative TB patient cost surveys for the Democratic Republic of the Congo and the Lao People's Democratic Republic were carried out from 2018–2019 *(89,90)*. These facility-based, cross-sectional surveys of TB patients collected information on direct medical, direct non-medical and indirect costs of living with TB, as well as coping mechanisms. Results of the two surveys highlight the following economic impact of TB on patients:

- In the Democratic Republic of the Congo, employment of any form among people with TB dropped from 65% to 43% for people on TB treatment; this drop was more severe among people with TB working in the informal economy with the proportion of employment going from 45% to 31%. Overall, 23% of patients lost their jobs due to TB and 78% lost working days (89).
- In the Lao People's Democratic Republic, the proportion of people with TB who became unemployed more than doubled during TB treatment (from 16.8% to 35.4%), while the proportion of employment in the informal economy decreased from 61.9% to 43.6%. The proportion of formal employment decreased from 10.8% to 8.8% and a very low level of benefits claimed (including sickness and unemployment benefits) was noted (90).

3.3.1 Ensuring a non-discriminatory work environment

Labour legislation and employment policies aim to ensure that people affected by a disease such as TB have their rights at work protected and do not suffer discrimination.

International Labour Standards specify that TB infection and treatment should not be the cause for discrimination at work nor termination of employment. Importantly, these standards are meant to be ratified and included into national labour legislation. Enforcement will require strengthening the actors of the labour sector, in particular workers' and employers' organizations, and labour inspectorates. The following are relevant ILO standards:

- the overarching Discrimination (Employment and Occupation) Convention, 1958 (No. 111) (91);
- the Termination of Employment Convention, 1982 (No. 158) Article 6, according to which temporary absence from work because of illness or injury shall not constitute a valid reason for termination of employment (92); and
- the HIV and AIDS Recommendation, 2010 (No. 200) which includes specific references to TB and calls for ending discrimination and promotion of equality and opportunity in employment settings, mentioning that workers should benefit from programmes aiming at preventing specific risks of occupational transmission of HIV and related transmissible diseases, such as TB; and should have access to available diagnostic and treatment services without facing any discrimination in their employment (93).

While these policies are essential tools to prevent and minimize discrimination in the workplace, it is also important to realize that social protection itself can also result in stigmatization of and discrimination against beneficiaries. Welfare or social protection stigma is defined as "the negative socio-physiological consequences or 'psychic costs' of being on welfare" (94). People with TB may go through stigmatizing or self-stigmatizing experiences associated with TB and/or with being a beneficiary of a form of legal protection or benefit. Discrimination linked to both conditions can be mutually reinforcing and can discourage people affected by TB from claiming the support they are entitled to, both inside and outside the workplace. In these circumstances the provision of adequate legal support is essential to address any discriminatory act, the violation of human rights, or any breach of the social protection policy in place (see Box 3.5).



Box 3.5 Providing legal support to people with TB in Kyrgyzstan and Kazakhstan

In Kyrgyzstan and Kazakhstan, legal support is provided to people with TB and other vulnerable populations by 'street lawyers', who have been mentored by a trained and experienced lawyer who then provides ongoing support and mentoring through information and counselling. This legal support programme also trains representatives from vulnerable populations on how to identify and document cases of human rights violations and to provide support on common issues such as receipt/renewal of identification documents, registration of citizenship (for migrants), help in formulating claim statements to the court, assistance in cases of stigma and discrimination from health care workers, discriminatory issues with landlords or employers, and obtaining cash benefits for people affected by TB.

Source: How legal aid helps end TB and HIV [website]. Kyrgyzstan: United Nations Development Programme; 2021 (https://www.undp.org/kyrgyzstan/news/how-legal-aid-helps-end-tb-and-hiv).

3.3.2 Building safe and healthy workplaces

In high TB burden settings, TB has implications for business operations, including decreased productivity, absenteeism, high turnover and the risk of further transmission *(26)*. This negative impact can be mitigated in the workplace by programmes that comprehensively address TB prevention, diagnosis, treatment and care.

TB is particularly prevalent in certain occupational sectors, including in health workers and workers in the mining industry. The recognition of TB as an occupational disease is the first step for people to be entitled to any compensation, either by the employer or the social protection system. For example, in South Africa, TB is considered as an occupational disease for health workers in settings where the disease is most prevalent (95). This recognition also allows the prevention of specific risks of occupational transmission and enables workers with TB to seek timely diagnosis and appropriate treatment, and to receive adequate compensation. Importantly, beyond legal entitlement, benefits guaranteed through the occupational disease status should be made accessible, adequate and sensitive to the needs of workers with occupational TB. Box 3.6 provides examples of strategies to support building safe and healthy workplaces.

The Joint WHO-ILO-UNAIDS policy guidelines on improving health workers' access to HIV and TB prevention, treatment, care and support services (96) includes recommendations to develop specific policies that aim to:

- ensure priority access for health workers and their families to services for prevention, treatment, care and support; and
- prevent discrimination against health workers with TB and establish programmes for reasonable accommodation and compensation, including, as appropriate, paid leave, early retirement benefits and death benefits in the event of occupationally acquired disease.

Box 3.6 Examples of strategies to support building safe and healthy workplaces

TB programmes can support building safe and healthy workplaces for workers with TB by:

- A. advocating for legal recognition of TB as an occupational disease;
- **B.** collaborating with legislators to establish or improve laws that protect people with TB from stigma and discrimination and provide legal support to people affected by TB;
- **C.** engaging with companies, workers, labour inspectors and occupational health services to communicate adequately the value of engaging in TB care, and gain commitment from respective managements;
- D. ensuring that businesses and services that are already addressing HIV expand to also address TB, and vice versa; and
- E. negotiating for the expansion of well-functioning TB and TB/HIV workplace programmes in small and medium-sized enterprises and other formal and informal workplaces with the support of civil society and relevant partners.

3.3.3 Granting income security through universal social protection

Universal social protection encompasses effective access to health care without hardship, recognizing that healthcare costs and health status impact household income (via reduced productivity due to poor health status, impoverishment due to healthcare costs, etc). Numerous countries, such as Rwanda, have taken steps towards integrating TB-related services into national social health protection programmes, with varying degrees of success (*26*).

An ILO review of selected high TB burden countries showed that treatment for TB and DR-TB was very often integrated into the design of benefit packages. The level of inclusion of other services, particularly for TB prevention, varied widely across countries and programmes. Furthermore, costs associated with travelling to health providers, as well as costs of ancillary drugs, food supplements or specific examinations, are often not included in benefit packages (26).

Income security in case of sickness in many countries has been established as the employer's liability (whereby the employer is required to continue paying wages while the employee is unable to work) rather than as a proper social protection programme. This creates a disincentive for employers to employ or retain workers affected by TB, as they may not be able to afford to retain workers who are unable to work for extended periods of time. Similarly, this means that individuals who are self-employed and fall sick with TB must bear the costs of sickness alone. Sickness benefits anchored in law and granted by the social protection system are most effective in protecting all workers with TB.

While some countries have implemented universal sickness benefits, globally the effective coverage of these programmes remains low due to gaps in compliance and implementation. Only a third of the world's working-age population have their income security protected by law in the event of sickness (15). For people with TB this means limited income protection when they are unable to work due to their health status or treatment, which results in loss or reduction of their source of income and possible loss of livelihoods.

TB programmes can contribute to protecting the income security of people affected by TB through several interventions as outlined in Box 3.7.

Box 3.7 Protecting income security of people affected by TB

TB programmes can contribute to protecting the income security of people affected by TB through several means, including by:

A.

No discrimination

Safe

Promoting a non-discriminatory social protection environment through:

- raising awareness among people with TB about their social protection rights;
- communicating positively the eligibility for social protection programmes or proposing non-stigmatizing targeting strategies to challenge the narrative according to which beneficiaries of social protection programmes are a burden for society because they are unwilling to work and/or are not sufficiently proactive in changing their socioeconomic status;
- providing people with TB and their households with clear information about social protection programmes, including their scope and the value of the benefits (such as enabling treatment completion or mitigating further impoverishment); and
- increasing awareness of TB, its risk factors and its clinical and financial consequences among staff working in social protection services.



B.

Increasing both the coverage and responsiveness of existing social protection benefits (including sickness and disability benefits) to address the needs of people with TB. Universal social protection C.

Advocating for the transition from the informal economy to the formal economy to ensure that all workers with TB have access to employment and social protection.

3.4 Monitoring the coverage of TB-sensitive social protection programmes among people affected by TB

Monitoring the performance of TB-sensitive social protection programmes primarily focuses on tracking the coverage and responsiveness of these programmes in addressing the needs of people affected by TB. A comprehensive monitoring and evaluation framework and standard indicators for social protection for people affected by TB are not yet available. This section therefore focuses on highlighting aspects that could be monitored during implementation of TB-sensitive social protection programmes.

Monitoring should focus on one or a very limited number of indicators related to the performance of social protection programmes in addressing the needs of people affected by TB. Examples of elements to be monitored include:

- the proportion of people diagnosed with TB whose social protection needs have been assessed
- the proportion of people diagnosed with TB who were referred to a social protection scheme; and
- The proportion of people affected by TB that is covered by any form of social protection.

In the political declaration of the 2023 High-level Meeting of the United Nations General Assembly (3), a new ambitious target was agreed to have 100% of people with TB accessing a health and social benefits package so they do not have to endure financial hardships because of their illness.

There are three main options for collecting (or obtaining) the data required to monitor TB-sensitive social protection programmes, as follows.

- 1. Link data held in the national TB database with national databases related to social protection programmes. If there are case-based digital databases for people diagnosed with TB and people enrolled in social protection programmes, and if these systems use unique identifiers that are common to both the health and social sectors, then an efficient option may be record-linkage between the two datasets (97). This option has the advantage of not requiring any additional collection of data. It does, however, rely on close collaboration between the national entities responsible for health and social protection. Analysis of data from such databases may allow disaggregation according to other relevant factors including age, gender, location, occupation and socio-economic status, and may thus enable monitoring of equity in access to social protection among people diagnosed with TB. This option of linking data held in different national databases is also highlighted in the latest WHO guidance on TB surveillance ([WHO Guidance on tuberculosis surveillance], [World Health Organization], in press, [2023]).
- 2. Collect data periodically, as part of other related studies. One of the three high-level targets of the WHO End TB Strategy (1) is to ensure that no TB patients or their households face catastrophic total costs as a result of TB disease. The recommended approach to measuring costs and assessing whether they are catastrophic is through nationally representative surveys of TB patients registered with the NTP (70). These surveys include in-depth interviews during which questions related to social protection could be asked.
- 3. Collect data as part of the routine TB surveillance system. This option is only feasible in countries with a case-based digital surveillance system. It is also important to consider whether such data collection is a good use of staff time and other resources. Periodic surveys or linkages between databases (as mentioned above) should be more efficient and less data collection intensive.





Credit: WFP/Francis Thawani

WHO recommends the provision of TB-specific social protection to individuals who have been diagnosed with TB to support them throughout the course of treatment *(29)*. This recommendation is based on established evidence on the benefit of material support including meals, food baskets, food supplements, food vouchers, transport subsidies (such as reimbursements, tokens, vouchers or passes), living allowances, housing incentives or financial bonuses for people with TB *(98)*. This list is not exhaustive and could be complemented by other forms of material support that contribute to improve the living conditions of people with TB and their households, if not adequately provided by the national social protection system. However, this section will focus mainly on the implementation of TB-specific social protection programmes in the form of socioeconomic and nutritional support.

Emerging evidence further suggests that some programmes for reducing social exclusion due to TB-associated stigma can be feasible and effective in high TB burden settings *(99)*. Stigma reduction programmes have the potential to improve outcomes for people affected by TB, but there is currently limited evidence on the appropriate intervention packages and their impact.

It is important to note that even when TB programmes wish to implement TB-specific social protection programmes, in particular when TB-sensitive programmes are either not available or not comprehensive enough, they should still assess whether it may be possible to run such schemes in a manner that benefits from the delivery infrastructure of the national social protection systems.

4.1 Socioeconomic support

WHO recommends TB-specific social protection programmes to support TB treatment adherence and to reduce the impact that the disease and its treatment have on the quality of life of people with TB (29). These programmes are also recommended to reduce the risk of people with TB and their households incurring catastrophic total costs, and to mitigate the impoverishment effect of TB (29).

This section focuses on the provision of TB-specific social protection programmes in the form of socioeconomic support, such as cash or vouchers. Box 4.1 outlines steps for implementing TB-specific social protection programmes based on socioeconomic support. The design and implementation of these programmes should be people-centred and based on an assessment of the needs, preferences and values of the beneficiaries (100). Consulting people affected by TB, healthcare providers and policy-makers can shape the intervention around community needs and help the assessment of its acceptability while maximizing its impact.



Box 4.1 Key considerations for implementing TB-specific social protection programmes based on socioeconomic support

The following key aspects should be taken into consideration when implementing TB-specific social protection programmes based on socioeconomic support.

- Determine the best location and strategy for the delivery of the social protection benefits and services Social protection and TB services can be delivered at the same facility depending on the specific benefits and services provided and the local administrative arrangements, following a so-called single-window approach. The adoption of digital tools, such as the transfer of cash benefits through mobile phones, can help reduce further financial strain on people affected by TB while expanding the coverage of social protection. Regardless of the strategy, it is important for the delivery of transfers to be reliable and predictable for the intended beneficiaries.
- 2 Determine the appropriate size and type of cash benefit, as well as the target populations The level of cash benefits may be informed by TB patient costs surveys, as well as national data on household expenditures, national poverty lines and wages, and can be adapted to respond to vulnerabilities determined in the social protection needs assessment.
- 3 Assess whether any additional support is appropriate within a "cash-plus" framework The cash-plus model is an approach for delivering social assistance whereby cash benefits are combined with another form of social support (such as in-kind transfers or services) when income support is not deemed sufficient to reduce households' vulnerabilities, including those that are health-related (101-103). This "plus" component or "intensifier" could be a top-up of the cash benefit to account for specific costs incurred in a particular setting, or for a specific category of people with TB (such as people with multidrug-resistant tuberculosis (MDR-TB) or people affected by TB and other comorbidities), or it could be the provision of a food basket to improve nutritional outcomes and/or the provision of psychosocial support to further reinforce the retention of patients in TB care.

4.2 Nutrition support

The association between social protection, food insecurity, nutrition and TB is well established (104-106). Most of the incident TB globally can be attributed to undernourishment (106,107). People with TB who are undernourished are also at increased risk of poorer TB treatment outcomes, higher mortality rates and a lower speed of recovery (104,108,109). WHO considers the management of malnutrition/undernutrition through nutrition support a key intervention when managing people with TB, and recommends the inclusion of nutrition support as part of a standard treatment and care plan for TB (98).

Food security is a complex state consisting of four different dimensions: physical availability of food, economic and physical access to food, food utilization (ability of food to provide adequate energy and nutrient intake) and the stability of these dimensions over time. Both transitory and chronic food insecurity may be a cause or a consequence of TB (110). Food insecurity is also closely related to diabetes, a key driver of the TB epidemic in some settings, and often a comorbidity of TB (111). Food insecurity has been associated with poor mental health and substance use, both known risk factors for poor TB treatment outcomes (112,113).

Programmes to address undernutrition can either involve direct nutrition support, such as provision of nutritional supplements or food baskets, or can take the form of vouchers or other transfers of cash intended to be used for food *(98)*. A combination of cash and food benefits may allow for a more flexible delivery and usage of the benefits. The direct provision of nutrition support has been found to improve TB treatment outcomes and adherence, and to reduce loss to follow-up in several low- and middle-income settings *(114-117)*.

WHO *Guidelines on nutritional care and support for patients with tuberculosis* and the WHO *Framework for Collaborative Action on Tuberculosis and Comorbidities* provide further and more detailed guidance on ways to address undernourishment in people affected by TB, in collaboration with other agencies implementing social protection (79,118). Box 4.2 lists some of the key considerations for the design and implementation of TB-specific social protection programmes based on nutrition support.

Box 4.2 Considerations for implementing TB-specific social protection programmes based on nutrition support

The following are key considerations for the design and implementation of TB-specific social protection programmes based on nutrition support.

- Determine the prevalence and severity of undernutrition in people with TB Assessment of individualspecific food and nutrition needs is best undertaken at the primary care level. According to existing WHO and WFP guidelines on nutritional support (7,119) key variables useful to monitor nutrition status of people with TB include:
 - BMI in adults over the 18 years of age;
 - BMI-for-age and BMI-for-sex z-score for children and adolescents aged 5–19 years; and
 - weight-for-length or weight-for-height z-score for children who are less than 5 years of age, with midupper arm circumference being used to identify cases in need of life saving nutrition management.

These variables can be recorded at the time of diagnosis and monthly until completion of treatment. Data on nutrition status is essential to prioritize the beneficiaries and packages for nutritional support *(120)*. Different delivery mechanisms and different combinations of direct food support and cash, for example, may be used. Considering the overlap in vulnerability to TB and undernutrition, collaboration with government and food support organizations (such as WFP or local NGOs) has synergistic benefits, as it allows the establishment of bilateral screening programmes whereby food support organizations can refer their clients for TB screening,



and TB programmes can refer people with TB to food support organizations for nutrition status screening. This approach may enable quantification of the extent to which the population entitled to food support matches with the people affected by TB, and thus can also inform co-programming decisions.

- 2. Use participatory methods to understand nutritional needs and the agriculture landscape. Programmes for addressing food insecurity and undernutrition can be more effective when cash or vouchers are combined with local nutrition support. Consultation with key stakeholders across sectors, including with affected communities and civil society, can inform the needs and barriers to accessing food (121). Specialized agencies can assist with market-based economic analysis to establish trends in prices, availability and seasonality, to inform decision making on nutritional support to people affected by TB. Establishing partnerships with multilateral organizations and local NGOs can be also helpful to explore economic and seasonality factors that may affect the nutritional needs of people affected by TB.
- 3. Assess the appropriate size and type of benefit The appropriate size and type of benefit provided can be determined from various sources, including national TB patient cost surveys, national data on household expenditure and poverty lines, and by using the WFP's Omega Value tool (122). This tool was developed by WFP to allow policy-makers to compare direct food transfers, food vouchers and cash transfers to maximize value for money for better programme results, impact on beneficiaries and cost savings (122). Providing cash in addition to food can also account for spending on commodities like soap for hand washing and fuel for cooking, both essential to ensure food biosafety, a critical determinant of nutrition. WFP and the Food and Agriculture Organization of the UN (FAO) indices of food prices can be used to determine amounts of staple foods to provide. Nutrition experts can help to determine the adequacy of the food mix needed to address the nutritional needs of people with TB.
- 4. Choose locally appropriate methods to deliver food support and appropriate communication material Food support can be delivered when people with TB are started on treatment, through the support of community health workers and/or the provision of electronic vouchers that households affected by TB can use to purchase food from the local markets. Providing nutritional support at the point of TB care may be more people-centred and convenient for the beneficiaries. A food support programme in East Timor was found to have no impact on treatment outcomes because the food was delivered at the clinic in the afternoon, requiring beneficiaries to forgo their work commitments to receive support (123).

People affected by TB should be informed of eligibility criteria, benefits, and delivery options and how to make the most of the benefits received to maximize nutritional impact.

5. Strengthen procurement options through collaboration with local actors – Collaboration with providers of available and affordable foods must be undertaken at all levels of the health system where nutritional support is delivered (for example, as part of TB clinics). Further, establishing procurement agreements with the private sector (including market owners) and reaching out to smallholder farmers is encouraged to strengthen synergistic development and ensure sustainable food supplies across seasons.

4.3 Interventions to reduce TB-associated stigma

Stigma and associated discrimination and social exclusion are major contributors to the vulnerability of people affected by TB. They may significantly increase costs incurred by people with TB, for example when people with TB are evicted from their homes, or when they lose their jobs.

While the effects of stigma resulting from TB are widely reported, there is a dearth of quality evidence to inform the development of guidelines on interventions to reduce TB-associated stigma *(99)*. Some of the best practices and tools presented in this section are based on experiences in several countries, as the following examples illustrate.

Examples of interventions with potential to reduce self-stigma

TB PhotoVoice (https://tbphotovoice.org) (124) – TB PhotoVoice is both an intervention and a participatory research method. It empowers people affected by TB to express and communicate their experiences through photography. Photography is a fast, easy and relatively inexpensive tool to explore and document the manifestations and impact of TB stigma (whether social, emotional or economic) on the individual, their families and the wider community. The material produced contributes to advocacy, helps sensitize key audiences such as decision makers, civil society and health staff, and allows a form of peer support for other people with TB, for example through media campaigns, exhibitions, cultural events and outreach to schools and women's groups.

From the Inside Out: Dealing with TB-related self-stigma and shame (125) – This training toolkit was developed by Beyond Stigma and KNCV Tuberculosis Foundation to assist people to identify, understand and address self-stigma and anticipated stigma related to TB. It is intended for use by national programmes, NGOs and TB support groups, and requires adaptation and testing in the local context.

Example of intervention to reduce TB-associated stigma among healthcare providers

The Allies Approach (126) – The Allies Approach, also developed by the KNCV Tuberculosis Foundation, is an intervention package aimed at font-line health workers and healthcare facilities, focusing on TB stigma at the emotional, cognitive, and practical level. It aims to address self-stigma in healthcare workers providing TB services, along with stigmatizing behaviour of healthcare workers and the policies in healthcare facilities that may contribute to stigma. The Allies Approach is designed to foster a dynamic and mutually supportive alliance between people with TB and care providers.

Examples of interventions to reduce community TB-associated stigma

The Right to Breathe (127) – This training tool was developed to address TB-related stigma in communities. It provides training to enhance community awareness and sustainable engagement in rights-based TB responses. The training manual, titled *The Right to Breathe: Human Rights Training for TB Survivors and Affected Communities,* aims to strengthen the capacity of people affected by TB in seeking accountability from governments and other duty bearers for their rights in the implementation of national TB strategies.

Understanding and challenging TB stigma: Toolkit for action (128) – This intervention is another tool to directly address stigma in the community. It was developed by and for trainers and can help trainers plan and organize participatory educational sessions with community leaders or organized groups to raise awareness and promote practical action to challenge HIV and TB-associated stigma and discrimination. It includes a range of participatory games, exercises, and picture-tools to help address TB-related stigma, suitable for a range of contexts and settings.

In addition, the products of **TB PhotoVoice**, made by people with TB expressing their experiences, as mentioned above, could support interventions to increase awareness of stigma in the community. There are currently no WHO recommendations on interventions to address TB-related stigma, and the above interventions implemented as operational research will contribute evidence to inform future WHO guidelines.



When such interventions are implemented, it is recommended to assess their feasibility and effectiveness under operational research conditions to inform national and global policies. If needed, interventions may be adapted to better fit into the local context and to best respond to local needs. Relevant local procedures for research, including obtaining ethical approval, should always be followed when implementing actions to address TB-associated stigma under operational research conditions, to generate solid evidence of impact to inform scale-up of the initiatives.

Box 4.3 provides a checklist of the key steps to follow when planning and implementing interventions to address TB-associated stigma.

Box 4.3 Steps for developing and evaluating interventions to address TB-related stigma



Create a multidisciplinary team and seek buy-in from stakeholders:

- map and engage relevant stakeholders, including civil society and affected communities;
- establish a multidisciplinary working group on TB-related stigma, which includes affected communities and other key stakeholders; and
- undertake training and capacity building on the assessment of TB-associated stigma.



Conduct a situation analysis of existing TB-related stigma:

- adapt and use stigma measurement tools for baseline assessments, such as those developed by the Stop TB Partnership and the KNCV Tuberculosis Foundation (129,130);
- estimate the burden of TB-associated stigma and its impact on access to TB care and social protection services;
- · determine the extent to which existing laws protect people from TB-associated stigma; and
- following this assessment compile a list of locally appropriate, suitable interventions.



Plan and mobilize resources for interventions to reduce TB-related stigma:

- disseminate the results of baseline assessments, sensitizing stakeholders on TB-related stigma;
- develop a roadmap and costed action plan, including operational research resources, to reduce TB stigma with the direct engagement of communities affected by TB;
- develop a framework to measure progress against the roadmap; and
- identify and mobilize resources for implementation of the roadmap.

Monitor and evaluate progress of the TB stigma reduction interventions:

- organize periodic stakeholder meetings to review progress in implementation of the selected intervention to address TB-associated stigma in order to identify and address challenges; and
- assess progress on key indicators for each intervention to build evidence on its effectiveness, feasibility and acceptability.



Develop manuscripts for peer-reviewed literature:

 disseminate methodological and operational lessons as well as impact findings emerging from the interventions.

4.4 Monitoring the performance of TB-specific social protection programmes

The monitoring and evaluation of TB-specific social protection programmes is useful to allow their adaptation and refinement so as to optimize impact. As for TB-sensitive social protection programmes, a comprehensive monitoring and evaluation framework and standard indicators for TB-specific social protection for people affected by TB are not yet available. This section therefore focuses on highlighting aspects that could be monitored during implementation of TB-specific social protection programmes, including:

- coverage of benefits under the TB-specific intervention, which can be disaggregated by relevant sociodemographic indicators (age, gender, race, disability, geography and socioeconomic position) to monitor coverage and equity; In line with the latest 2023 political declaration of the High-level Meeting of the United Nations General Assembly (3), countries could set a provisional target of 100% coverage among those eligible for the proposed social protection benefits;
- the amount of cash benefits or food support received and duration of enrolment in the programme;
- the proportion of cash support spent on food, transport, additional TB care and other commodities;
- the proportion of households affected by TB classified as poor prior to and after the implementation of the intervention.¹
- the proportion of households affected by TB classified as food insecure (as measured through an established indicator of food insecurity such as the ones consolidated by the WFP) (132) prior to and after the implementation of the programme;
- the proportion of households affected by TB already covered by any other form of social protection programme (on top of the TB-specific one) and/or receiving any form of food supplementation or support at the start of TB treatment;
- the proportion of people with TB who completed TB treatment and were cured, stratified by beneficiary status (beneficiaries of the programme vs non beneficiaries);
- the proportion of people with TB who died, stratified by beneficiary status (beneficiaries of the programme vs non beneficiaries); and
- the proportion of households affected by TB who incurred catastrophic total costs, stratified by beneficiary status (beneficiaries of the programme vs non beneficiaries).

¹ The extent of poverty rates changes pre- and post-TB-disease are usually evaluated through quantitative or qualitative methods. Quantitative methods used in national TB patient cost surveys evaluate pre- and post-TB-disease household poverty rates by comparing respondent's daily income against the international poverty threshold *(80)*. In addition, perceived impoverishment (qualitative method) is also enquired systematically as part of the social impact of TB episode questions *(78,131)*.

Establishing and social protection



TB programmes have a crucial role to play in promoting the inclusion of people affected by TB within existing social protection programmes and in the delivery of TB-specific social protection programmes. The integration of TB and social protection programmes is part of the multisectoral response to end TB, with actions by different sectors and stakeholders to holistically address the determinants of TB. The WHO MAF-TB *(8)* provides a mechanism through which this multisectoral response can be coordinated, monitored and reviewed. MAF-TB is aligned with the HiAP approach, which recognizes that health is not merely a responsibility of health sectors programmes but can be profoundly influenced by policies that guide actions beyond the health sector.

The HiAP literature assembles information on collaborative practices associated with positive experiences leading to health and improvements in health equity through addressing the social determinants of health (133). Several of the collaborative practices which are important to sustain intersectoral work are included in the MAF-TB (8). Additionally, the HiAP provides a generic assessment framework for understanding the important processes and functions needed to sustain collaboration between sectors with different goals, mandates and hierarchies (134).

This section provides an overview of the MAF-TB (8) and HiAP frameworks and how they can serve the establishment of effective multisectoral collaborations. It also includes a description of the operational features that characterize successful multisectoral collaborations building upon the HiAP approach.

5.1 Multisectoral collaboration for action on the social determinants of TB: challenges and opportunities

Multisectoral collaboration has been defined as "a recognized relationship between part or parts of the health sector with part or parts of another sector which has been formed to take action on an issue to achieve health outcomes (or intermediate health outcomes) in a way that is more effective, efficient, or sustainable than could be achieved by the health sector acting alone" (135).

There is a strong rationale for closer integration of TB and social protection services. However, many barriers may hamper the full establishment and consolidation of effective collaboration between government ministries. For example, it may be difficult to achieve a common understanding of the issues, to define shared goals and to agree on criteria for success. Collaboration may need to be sustained for years to bring about the meaningful social change required to address the needs of people with TB, and this may require continuous and adequate funding and incentives. Government and social and healthcare workers need supportive, stable, and accountable relationships across their institutions, which because of their respective mandates, may not always prioritize the needs of people with TB.

Despite these challenges, opportunities and examples exist of successful multisectoral collaboration across the TB and social protection sectors. The Brazilian *Bolsa de Familia* (family cash benefits) programme – now called *Auxilio Brasil* – is a conditional social protection programme in which patients undergoing treatment for TB and their households have been included. In this context the municipal government plays an important role to ensure people with TB are provided with food (particularly breakfast) in the establishments where they receive TB treatment. Similarly, in Colombia, the identification of different institutions and entities for social protection which could be used for empowerment of the TB programme in the neighbourhood of *Rafael Uribe* in Bogotá was a function the local governments performed *(136)*. These implementation features at the local level mean that within TB and social protection collaboration, there is an important role for local authorities, whose line ministries are often different to those for social protection and health. A further important aspect of developing and maintaining multisectoral collaboration for furthering social protection coverage of people with TB is the focus on the integration of health and social services, as Colombia's experience showed *(136)*. Good referral mechanisms are required across different social services to healthcare services, and vice versa. Again, this means



that multisectoral collaboration mechanisms need to embed participation of key service-level representatives from both social care and healthcare sectors and pay attention to the ways of working and resourcing that foster this collaboration.

5.2. The multisectoral accountability framework to accelerate progress to end TB

The WHO MAF-TB (8) aims to support effective collaboration within and beyond the health sector, and to support accountability of governments and all stakeholders at global, regional and country levels in order to accelerate progress to end the TB epidemic in line with the *End TB Strategy* (1) and the 2030 Agenda for Sustainable Development (6). The framework aims to establish the conceptual and operational foundation to foster multisectoral action, mutual accountability and measurement of progress towards *End TB Strategy* targets. It is a practical tool through which countries can establish and monitor multisectoral collaborations, and is comprised of four components: commitments, actions, monitoring and reporting, and review (137).

MAF-TB was developed in response to the acknowledgment that ending TB requires coordination of multiple sectors able to tackle the social and structural determinants of the epidemic. Engagement with the social protection sector and with social protection programmes for people affected by TB falls under the "actions" component of MAF-TB (8).

In fact, implementation of social protection programmes for people affected by TB represents one of the best examples of multisectoral response to TB. As such, social protection can be one of the most obvious areas through which countries can monitor progress and success of multisectoral collaboration to end of TB.

To facilitate implementation of MAF-TB (8) at country level, WHO developed the MAF-TB checklist and supported baseline assessments (137). Results from the assessments inform further steps to adapt and implement MAF-TB at national level. Findings from the related baseline assessments conducted in 2021 on a sample of 45 countries (138,139) suggest that countries are all making progress in the implementation, monitoring and evaluation of multisectoral collaboration. To further support countries in MAF-TB operationalization, WHO developed two documents – an operational guide and a compilation of best practices on MAF-TB adaptation and implementation (139,140). The operational guide provides practical advice on key approaches and interventions needed to establish the MAF-TB at the national and local levels, with concrete country examples, best practices and case studies under each suggested approach and intervention (140). The compilation of best practices includes 25 case studies from all six WHO regions. It provides important, valuable insight into how regions and countries are progressing with MAF-TB implementation and what has been learned along the way (139).

Since 2020 global data has been collected annually to track progress in MAF-TB adaptation and implementation in countries from all WHO regions (11). It is important to continue collecting data and in particular, more concrete examples of collaborative models between the TB and other sectors, in order to consolidate the lessons learned and maximize their generalizability to other contexts.

5.3 Establishing effective multisectoral collaboration: the Health in All Policies (HiAP) approach

Multi-, cross- or inter-sectoral approaches such as HiAP (see Box 5.1) are increasingly acknowledged as effective and evidence-based collaboration and governance strategies to tackle the social determinants of health for advancing health equity (141,142).

Box 5.1 Definition of the HiAP approach

HiAP is an approach to public policies across sectors that systematically takes into account the health implications of decisions, seeks synergies, and avoids harmful health impacts in order to improve population health and health equity.

HiAP is founded on health-related rights and obligations and contributes to strengthening the accountability of policy-makers for health impacts at all levels of policy-making. It emphasizes the consequences of public policies on health systems, determinants of health, and wellbeing. It also contributes to sustainable development.

It is recognized that governments are faced with a range of priorities and that health and equity may not automatically gain precedence over other policy objectives. Nonetheless, health considerations do need to be taken into account in policy-making. Efforts must be made to capitalize on opportunities for co-benefits across sectors and for society at large. Effective safeguards to protect policies from distortion by commercial and vested interests and influence also need to be established (134).

Source: Health in all policies: Helsinki statement. Framework for country action. Geneva: World Health Organization; 2014 (https://www.who.int/publications/i/item/9789241506908).

The social determinants of health are the conditions in which people are born, grow, work, live and age, and the wider set of forces and systems shaping the conditions of daily life (143). Social determinants of health have an important influence on health inequities – the unfair and avoidable differences in health status seen within and between countries. In countries at all levels of income, health and illness follow a social gradient: the lower the socioeconomic position, the worse the health (144). As discussed in Section 1, TB is typically characterized by deep inequalities with low socioeconomic position groups and marginalized segments of the population bearing the heaviest burden of TB.

Social protection is one of the key policy domains to improve the social determinants of health, as well as addressing social inequality and human rights, and is protective for TB and other health conditions. While examples of multisectoral collaboration in TB already exist, new and innovative mechanisms for multisectoral collaboration may be especially needed to contrast emerging and potentially more dangerous variants of infectious pathogens, like MDR-TB. These resistant strains in particular have the potential to threaten global health and financial security and require coordinated efforts from multiple sectors to anticipate and mitigate their damaging effect.

Fig. 5.1 synthesizes important features underpinning effective multisectoral collaboration into a single HiAP model. These features are generically valid across different types of multisectoral work. The HiAP model provides a roadmap and checklist for the set-up of multisectoral work for health and social protection, giving ideas of what areas may need strengthening to improve performance of the collaboration. Notably, the HiAP approach requires working across different sectors but also across hierarchies within sectors, which are inherent in governmental, political, administrative and judicial systems. Across sectors and levels, the HiAP model organizes knowledge



on effective multisectoral collaboration into three main components: "foundations" representing values and principles; "arches" describing the context of determinants of health; and "pillars" which represent core features of the organizational and institutional environments ([Sustainable multisectoral collaboration to address the social determinants of health, equity and well-being. Practical guidance based on health-in-all policies approaches], [World Health Organization], unpublished, [2023]).





Source: Working together for equity and healthier populations: sustainable multisectoral collaboration based on Health in All Policies approaches. Geneva: World Health Organization; 2023. Licence: CC BY-NC-SA 3.0 IGO. Page 21 https://www.who.int/publications/i/item/9789240067530.

5.3.1 Values and principles: the foundations

The foundations of the HiAP model bring together the values and principles that regulate collaborative efforts across different sectors. These values and principles focus on the public value of collaborations that promote health, health equity and sustainability, and align well with some of the equivalent principles outlined in the *Guidance for national strategic planning for tuberculosis (69)*. Building upon this alignment, Table 5.1 gives examples that illustrate the concrete application of these principles and values in establishing strong and successful collaborations between the TB and social protection sectors. While these examples are useful to undertake concrete actions, it should be acknowledged that much of the success of multisectoral engagements starts with the mindsets of the stakeholders involved: both parties involved should commit to these principles and recognize the added value of and need for the multisectoral collaboration proposed.

Table 5.1 Key principles of successful multisectoral collaborations: concrete examples for TB and social protection

General principles	Areas of application for successful collaborations between the TB and social protection sectors
Building trust and relationships	 Establishing high-level government leadership and oversight of the social protection planning and implementation stages: both stages should be overseen by the highest possible government office hierarchically above the TB programme. This level of leadership provides the administrative authority to engage key decision-makers in other ministry of health departments and in other government ministries and agencies, as well as stakeholders beyond the health sector. Engagement of all key stakeholders throughout the planning, implementation and evaluation of social protection programmes for people affected by TB: this engagement should give stakeholders opportunities to influence the course of the planning process and to be embedded in the multidisciplinary and multisectoral team (see Section 2) where possible.
Respectful and responsive to partners' needs	• Understanding of the mandates and perspectives of the partners and stakeholders to establish respectful and sustainable collaborations: there are often areas where TB programmes can meaningfully engage and contribute to the achievement of the mandates of the social protection sector, to create a symbiotic relationship with reciprocal benefits. This requires the provision of basic information and data to demonstrate the role of the social protection sector in the TB response, but also how a more TB-sensitive approach can be beneficial to best meet the goals of social protection programmes. For example, TB programmes may want to leverage the poverty alleviation mandate of social protection implementation agencies by presenting evidence on the impoverishment effect of TB. Also, as TB is an important tracer of health inequities, both declining TB rates and improving treatment outcomes are important signals of the capacity of existing social protection systems to be progressive and equity oriented.
Flexibility and adaptability	 Building multisectoral collaborations using existing collaborative platforms: engagement with social protection stakeholders should be done in a manner that does not create administrative or financial overload for them. Successful collaborations typically adapt and respond flexibly to existing resources and opportunities rather than creating new areas of work or investment – an excellent example of this is the rapid mobilization of health and social protection resources in response to the COVID-19 pandemic. TB programmes may follow this example by undertaking a comprehensive mapping of resources in order to explore available resources, propose more strategic and efficient co-funding mechanisms, and advocate for better financial resources to support a truly multisectoral TB response.
Transparent, open communication	 Engaging in dialogue with stakeholders as equals – valuing their knowledge and perspectives: stakeholders should be given the opportunity to have internal consultations and help to determine their own engagement in the planning and implementation process of social protection programmes for people affected by TB. Ensuring that communications can be easily understood by all stakeholders: this can be achieved, for example, by avoiding complex or technical language during stakeholder engagement.
Build a skilled HiAP workforce	• Developing educational material and joint training programmes involving TB programmes and social protection staff: this step is important to foster a common vision and shared understanding of each other's domains, and to reinforce a mutual sense of responsibility.



5.3.2 The context of determinants of health: the arches

The arches show the broad contextual factors that shape and influence health and well-being. The first arch, which comprises social, political, environmental, cultural, commercial and economic determinants of health, reflects the powerful, comprehensive influence of determinants on health and well-being, as well as on social equity. The second arch, which comprises action on the structural determinants of health equity, highlights the importance of considering structural drivers of social equity that arise from different systems. Structural drivers include the distribution of power, money and resources and the role those forces play at all levels of society, including government, institutions and communities, in changing social equity and the burden of disease. The third arch – moving towards HiAP sustainability – acknowledges that it takes time to embed multisectoral collaboration so that it survives changing political and bureaucratic cycles. As confidence in the value of cross-sectoral collaboration increases, governments and services will be motivated to invest more to strengthen multisectoral work as part of daily business. The fourth and final arch – culture of collaboration and integration – is the label used to describe the four pillars of the HiAP model, as described below.

5.3.3 Core features of the organizational and institutional environments: the pillars

The four pillars at the heart of HiAP are: (i) governance and accountability, (ii) leadership at all levels, (iii) ways of working (working methods) and (iv) resources, financing and capabilities. All four pillars are heavily dependent on the involvement of concerned people from both the health and social protection sectors, for the implementation and monitoring of social protection policies relevant to public health. The ambition of the four pillars is to sustain an organizational and societal culture in which multisectoral collaboration thrives, to enable health and social protection policies and services to be comprehensive and aligned to populations needs. The 11 key action areas related to each pillar describe these ambitions in more detail and are further illustrated in the concrete examples below.

An authorizing environment and mandate to act

A mandate for HiAP from the highest level of government is needed because cross-sectoral collaboration operates outside of the normal hierarchy of command. This is consistent with one of the most important actions under MAF-TB (8), which demands the establishment of a multisectoral multistakeholder coordination and review mechanism under the highest possible leadership level. Authorizing governance instruments for multisectoral collaboration may include intercountry or presidential decrees or directives, enabling legislation or development strategies.

Example: An authorizing mandate to work multisectorally to address TB in mining worker populations came to Southern African governments from a supra-governmental regional gathering (145). Nonetheless, without translation into national mandates and collaborative structures, this opportunity has not delivered promised improvements for TB through improved social protection programmes (145,146).

Layered cross-government committees to support collaborations

Multisectoral governance structures provide ongoing authority and high-level executive oversight. Horizontal governance (between sectors, such as via memoranda of understanding) facilitates operational levels of policy-making and projects. Multi-level mechanisms facilitate coherent policy-making from the national level down to local levels, to ensure the realities at local level are addressed. Some government ministries with less sectoral focuses, such as ministries for social development for example, may be useful facilitators of multisectoral structures.

Example: The role of the Ministry of Social Development in early child development programmes in Chile showed how collaborative structures could facilitate delivery of multisectoral services on the ground (147).

Whole-of-government plan for policy action

A whole-of-government plan with a clear shared vision and objectives is a key vehicle to support collaboration and can drive investments in expanded services and coverage. Collaborations between health and social protection may connect to existing government plans or sponsor broader health development and well-being plans that then involve many more sectors. Well-conceived plans usually contain targets and obligations for measuring progress.

Example: The San Juan de Lurigancho Health Network in Lima, Peru, is an example of how a common evaluation was undertaken of the multiple activities and commitments made by the public and private institutions and organizations that formed the High Committee for the TB network. Having a transparent plan and evaluation improved accountability and facilitated making a strong case for continuing the work (148).

Support for collaboration on shared policies, projects and proposals

Shared policy or project proposals should be drawn up, outlining collaborative opportunities and responsibilities for improving coverage of social protection for people affected by TB. It is essential to establish a common understanding of the problem of TB early on in the collaboration, and the role that social protection can play in addressing it. This allows sectoral stakeholders to explore their positions, values and experiences such that joint service, project or policy proposals are fully owned by individual sectors. Leadership at all levels is important for these discussions, to facilitate coherent, cohesive and shared goals. A key part of this process will be to ensure that proposals are endorsed at both the highest level of government and by each agency or ministry. Updates on progress should be provided regularly.

Example: The reorienting health programmes initiative developed in Chile with WHO (INNOV8) prepares health programme managers to analyze multidisciplinary data, including how social inequities reduce and limit access to health services, and how the programmes can be designed differently. Investments in understanding the social determinants of TB and the health and financial challenges experienced by people with TB should be prioritized. This is essential for the development of a shared vision of the problem, which ultimately is critical for informing co-designed solutions (149).

Advocating for HiAP and other collaborative approaches

One of the first elements of HiAP leadership at all levels is advocacy. It requires decision-makers and policy officers and practitioners to advocate for new HiAP activities and to enable collaboration to continue. Ongoing advocacy should be undertaken by leaders at all levels. Effective advocacy for HiAP requires people with skills in diplomacy and negotiation and the ability to navigate the political and policy imperatives of other agencies. These advocates are usually policy entrepreneurs who are able to seize windows of opportunity. Characteristics of effective HiAP advocates are that they can communicate and articulate the vision of the benefits from collaboration and win others over to embrace and implement it.

Example: The recommendation of Brazil's National Health Council (the highest advocacy forum for health) in 2011 advocated an all-of-society approach to support collaboration to ensure social protection for people with TB, as encapsulated by the successful Brazil programme, Bolsa Familia. It recommended "…for the Ministry of Health joined with other areas of the Federal Government, with the participation and support of social movements, the National Congress and institutions from other sectors, to create and maintain social benefits for people with TB, so as to increase treatment adherence and reduce TB incidence rate" (150).



Fostering a culture of collaboration and involvement of civil society and affected people

Leaders ensure the conditions necessary to help others to collaborate. Fostering a culture of collaboration requires including incentives for individuals to collaborate, such as personal growth and career development incentives, as well as valuing the principles of HiAP (see Table 5.1). Fostering collaboration that supports involvement of civil society and disadvantaged communities is an essential contributor to change based on a clear understanding of community needs. Involvement of academics and clinicians can ensure timely access to evidence and the inclusion of perspectives of people working in different areas of TB care and prevention.

Example: A Peruvian evaluation of a social and community collaboration network aiming at improving the control of TB in the country showed how having common TB-related objectives across different sectoral actors helped to bring the different participants to find collective solutions to adverse situations and to produce social innovations (151).

A network of HiAP champions

A HiAP champion is a person who takes an interest in and advocates for the adoption, implementation and success of a HiAP cause, policy, programme or project and is generally willing to work on innovations within organizations rather than stay within existing lines. Networks provide semi-formal or informal mechanisms for troubleshooting and give innovators inspiration and forward momentum. In multisectoral work with the social protection sector, HiAP champions from the social protection sector are among the most important leaders for change. They recognize the potential public value of expanding social protection for improved health outcomes. Networks can involve collections of actors who may not normally formally interact through traditional organizational platforms.

Example: Wales has established a Future Generations Commissioner to champion the delivery of multisectoral well-being goals outlined in its Wellbeing of Future Generations Act (152). As a further example, Finland draws on an extensive research network to inform HiAP action at various levels of government (153).

Developing collaborative partnerships built on trust

As HiAP is based on the concepts of mutuality and reciprocity, the nature of relationships and partnerships, from long-term partnerships to networks and informal exchanges, is crucial. Open, trusting relationships and communication hold partnerships together; without trust and open communication, collaboration is not viable.

Example: Fairer Healthier Scotland led to a collaborative approach to addressing health inequity, whereby public health staff engage directly with various levels of government, departments and sectors to influence policy and practice (154).

Using a co-design approach and understanding the priorities of other sectors

To work across sectors, policy officers must know and navigate their agency's interests and priorities and also understand the motivations and interests of other agencies. It is not unusual that this creates tensions within government, with the emergence of conflicts about values and diverging interests. Co-designing policies, projects and activities can promote understanding and trust to discuss and resolve issues as they arise and ensure a clear direction for policy.

Example: Urban regeneration programmes have been found to have potential impacts on the web of social relationships and social organization within communities, and in turn influence individuals' health seeking behaviour and other attitudes to TB risks, for example, in relation to stigma (9,155).

Dedicated HiAP roles and budget

Dedicated roles are important for HiAP activities and approaches. Allocation of sufficient, appropriate resources is important. Collaboration takes time and some financial resources are needed for purposes of collaborative work (such as for joint workshops or joint trainings). The funds could include an allocation from the ministry of health as well as all the other relevant ministries, to convene workshops, collect evidence, develop communication materials and reports, and establish HiAP training programmes.

Example: Vocational rehabilitation services (geared towards assisting individuals with chronic health problems to return to or remain in work) were tested in Sweden as a focus of co-financing. The co-financing approach trialled was for sectors responsible for social security, social welfare and health services to voluntarily pool up to 5% of their sectoral budgets through the formation of financing associations to manage these joint budgets. Qualitative evaluations, however, found a positive effect on coordinated care, and staff working in health centres reported better collaboration (156).

Capabilities to act on the determinants of health

Leaders identify and create opportunities to support others within their circle of influence to gain the confidence and skills needed to work collaboratively, through mentoring and other means for developing capability.

Example: In the Colombia TB Network, human resources capabilities is a strong focus, with emphasis on the transfer of knowledge between professionals from different entities and territories (157).

References

- 1. The end TB strategy. Geneva; 2015 (WHO/HTM/TB/2015.19; https://www.who.int/publications/i/item/WHO-HTM-TB-2015.19, accessed 31 August 2023).
- 2. Moscow Declaration to End TB. First WHO Global Ministerial Conference. Ending TB in the sustainable development era: a multisectoral response. Moscow: World Health Organization, Ministry of Health of the Russian Federation; 16-17 November 2017 (https://apps.who.int/iris/rest/bitstreams/1374353/retrieve, accessed 30 August 2023).
- **3.** Resolution A/RES/78/5. Political declaration of the UN General Assembly High-Level Meeting on the fight against tuberculosis. Resolution adopted by the General Assembly on 5 October 2023. New York: United Nations; 2023 (https://daccess-ods.un.org/access.nsf/Get?OpenAgent&DS=A/RES/78/5&Lang=E, accessed 6 March 2024).
- C102 Social Security (Minimum Standards) Convention, 1952 (No. 102). Convention concerning minimum standards of social security. Geneva: International Labour Organization; 1952 (https://www.ilo.org/dyn/normlex/en/f?p=NORMLEXPU B:12100:0::NO::P12100_INSTRUMENT_ID:312247, accessed 5 September 2023).
- R202 Social Protection Floors Recommendation, 2012 (No. 202). Recommendation concerning National Floors of Social Protection. Geneva: International Labour Organization; 2012 (https://www.ilo.org/dyn/normlex/en/f?p=NORMLEX PUB:12100:0::N0::P12100_INSTRUMENT_ID:3065524, accessed 5 September 2023).
- 6. Resolution 70/1. Transforming our world: the 2030 agenda for sustainable development. Resolution adopted by the General Assembly on 25 September 2015. New York: United Nations; 2015 (A/RES/70/1; https://www.un.org/en/development/desa/population/migration/generalassembly/docs/globalcompact/A_RES_70_1_E.pdf, accessed 5 September 2023).
- 7. Implementing the end TB strategy: the essentials, 2022 update. Geneva: World Health Organization; 2022 (https://apps.who.int/iris/rest/bitstreams/1486092/retrieve, accessed 30 August 2023).
- 8. Multisectoral accountability framework to accelerate progress to end tuberculosis by 2030. Geneva: World Health Organization; 2019 (https://apps.who.int/iris/rest/bitstreams/1276221/retrieve, accessed 30 August 2023).
- **9.** Hargreaves JR, Boccia D, Evans CA, Adato M, Petticrew M, Porter JD. The social determinants of tuberculosis: from evidence to action. Am J Public Health. 2011;101:654-62. doi: 10.2105/AJPH.2010.199505.
- Költringer FA, Annerstedt KS, Boccia D, Carter DJ, Rudgard WE. The social determinants of national tuberculosis incidence rates in 116 countries: a longitudinal ecological study between 2005-2015. BMC Public Health. 2023;23:337. doi: 10.1186/s12889-023-15213-w.
- 11. Global tuberculosis report 2022. Geneva: World Health Organization; 2022 (https://apps.who.int/iris/rest/bitstreams/1474924/retrieve, accessed 30 August 2023).
- **12.** Siroka A, Ponce NA, Lonnroth K. Association between spending on social protection and tuberculosis burden: a global analysis. Lancet Infect Dis. 2016;16:473-9. doi: 10.1016/S1473-3099(15)00401-6.
- 13. Colgrove J. The McKeown thesis: a historical controversy and its enduring influence. Am J Public Health. 2002;92:725-9. doi: 10.2105/ajph.92.5.725.
- 14. Words matter: suggested language and usage for tuberculosis communications. Geneva: Stop TB Partnership; 2022 (https://www.stoptb.org/file/10703/download, accessed 30 August 2023).
- **15.** World social protection report 2020–22: social protection at the crossroads in pursuit of a better future. Geneva: International Labour Organization; 2021 (https://www.ilo.org/wcmsp5/groups/public/---ed_protect/---soc_sec/documents/publication/wcms_817572.pdf, accessed 31 August 2023).
- **16.** The Universal Declaration of Human Rights (Resolution 217A (III)), 10 December 1948. New York: United Nations General Assembly; 1948 (http://www.un.org/en/universal-declaration-human-rights/, accessed 5 September, 2023).
- 17. Building social protection systems: international standards and human rights instruments. Geneva: International Labour Organization; 2021 (https://www.ilo.org/wcmsp5/groups/public/---ed_protect/---soc_sec/documents/publication/wcms_651219.pdf, accessed 31 August 2023).
- **18.** About the right to social security and human rights. OHCHR and the right to social security [website]. Office of the High Commissioner for Human Rights; 2023 (https://www.ohchr.org/en/social-security/about-right-social-security-and-human-rights, accessed 5 September 2023).
- **19.** Resolution concerning the second recurrent discussion on social protection (social security). Geneva, Switzerland: International Labour Organization; 2021 (https://www.ilo.org/wcmsp5/groups/public/---ed_norm/---relconf/ documents/meetingdocument/wcms_806099.pdf, accessed 31 August 2023).

- **20.** Ortiz I, Chowdhury A, Muzaffar T, Urban S. Fiscal space for social protection: a handbook for assessing financing options. Geneva: International Labour Organization; 2019 (https://www.ilo.org/wcmsp5/groups/public/---ed_protect/---soc_sec/documents/publication/wcms_727261.pdf, accessed 31 August 2023).
- **21.** Social security inquiry: manual. International Labour Office; 2016 (https://www.ilo.org/wcmsp5/groups/public/---dgreports/---integration/documents/instructionalmaterial/wcms_568711.pdf, accessed 31 August 2023).
- 22. Public Employment Programmes (PEPs): creating decent jobs through national investments. Geneva: International Labour Organization; 2020 (https://www.ilo.org/wcmsp5/groups/public/---ed_emp/documents/publication/wcms_759118.pdf, accessed 31 August 2023).
- **23.** Boccia D, Pedrazzoli D, Wingfield T, Jaramillo E, Lonnroth K, Lewis J et al. Towards cash transfer interventions for tuberculosis prevention, care and control: key operational challenges and research priorities. BMC Infect Dis. 2016;16:307. doi: 10.1186/s12879-016-1529-8.
- 24. Lutge E, Lewin S, Volmink J, Friedman I, Lombard C. Economic support to improve tuberculosis treatment outcomes in South Africa: a pragmatic cluster-randomized controlled trial. Trials. 2013;14:154. doi: 10.1186/1745-6215-14-154.
- **25.** Oliosi JGN, Reis-Santos B, Locatelli RL, Sales CMM, da Silva Filho WG, da Silva KC et al. Effect of the Bolsa Familia Programme on the outcome of tuberculosis treatment: a prospective cohort study. Lancet Glob Health. 2019;7:e219-e26. doi: 10.1016/S2214-109X(18)30478-9.
- **26.** Making universal social protection a reality for people living with, at the risk of, and affected by HIV or tuberculosis. Geneva: International Labour Organization; 2021 (https://www.ilo.org/wcmsp5/groups/public/---dgreports/---gender/ documents/publication/wcms_830750.pdf, accessed 31 August 2023).
- 27. Bono Joaquín Gallegos Lara beneficia a cerca de 23 mil cuidadores de personas con discapacidad severa [Joaquín Gallegos Lara voucher benefits close to 23 thousand caregivers of people with severe disabilities] [website]. Gobierno de la República del Ecuador [Government of Ecuador]; 2012 (https://www.presidencia.gob.ec/bono-joaquin-gallegos-lara-beneficia-a-cerca-de-23-mil-cuidadores-de-personas-con-discapacidad-severa/, accessed 6 September 2023).
- 28. Carter DJ, Glaziou P, Lonnroth K, Siroka A, Floyd K, Weil D et al. The impact of social protection and poverty elimination on global tuberculosis incidence: a statistical modelling analysis of Sustainable Development Goal 1. Lancet Glob Health. 2018;6:e514-e22. doi: 10.1016/S2214-109X(18)30195-5.
- **29.** WHO consolidated guidelines on tuberculosis. Module 4: treatment drug-resistant tuberculosis treatment. Geneva: World Health Organization; 2020 (https://apps.who.int/iris/rest/bitstreams/1280998/retrieve, accessed 31 August 2023).
- **30.** Aragao FBA, Arcencio RA, Fuentealba-Torres M, Carneiro TSG, Souza LLL, Alves YM et al. Impact of social protection programs on adults diagnosed with tuberculosis: systematic review. Rev Bras Enferm. 2021;74:e20190906. doi: 10.1590/0034-7167-2019-0906.
- **31.** Richterman A, Steer-Massaro J, Jarolimova J, Luong Nguyen LB, Werdenberg J, Ivers LC. Cash interventions to improve clinical outcomes for pulmonary tuberculosis: systematic review and meta-analysis. Bull World Health Organ. 2018;96:471-83. doi: 10.2471/BLT.18.208959.
- **32.** Andrade KVF, Nery JS, Souza RA, Pereira SM. Effects of social protection on tuberculosis treatment outcomes in low or middle-income and in high-burden countries: systematic review and meta-analysis. Cad Saude Publica. 2018;34:e00153116. doi: 10.1590/0102-311X00153116.
- **33.** Ciobanu A, Domente L, Soltan V, Bivol S, Severin L, Plesca V et al. Do incentives improve tuberculosis treatment outcomes in the Republic of Moldova? Public Health Action. 2014;4:S59-63. doi: 10.5588/pha.14.0047.
- **34.** Ukwaja KN, Alobu I, Gidado M, Onazi O, Oshi DC. Economic support intervention improves tuberculosis treatment outcomes in rural Nigeria. Int J Tuberc Lung Dis. 2017;21:564-70. doi: 10.5588/ijtld.16.0741.
- **35.** Ngamvithayapong-Yanai J, Luangjina S, Nedsuwan S, Kantipong P, Wongyai J, Ishikawa N. Engaging women volunteers of high socioeconomic status in supporting socioeconomically disadvantaged tuberculosis patients in Chiang Rai, Thailand. Western Pac Surveill Response J. 2013;4:34-8. doi: 10.5365/WPSAR.2012.3.4.013.
- **36.** Sripad A, Castedo J, Danford N, Zaha R, Freile C. Effects of Ecuador's national monetary incentive program on adherence to treatment for drug-resistant tuberculosis. Int J Tuberc Lung Dis. 2014;18:44-8. doi: 10.5588/ijtld.13.0253.
- **37.** Yao H, Wei X, Liu J, Zhao J, Hu D, Walley JD. Evaluating the effects of providing financial incentives to tuberculosis patients and health providers in China. Int J Tuberc Lung Dis. 2008;12:1166-72.
- **38.** Baluku JB, Nakazibwe B, Twinomugisha B, Najjuuko R, Isabella N, Nassozi S et al. One dollar incentive improves tuberculosis treatment outcomes in programmatic settings in rural Uganda. Sci Rep. 2021;11:19346. doi: 10.1038/s41598-021-98770-7.
- **39.** Torrens AW, Rasella D, Boccia D, Maciel EL, Nery JS, Olson ZD et al. Effectiveness of a conditional cash transfer programme on TB cure rate: a retrospective cohort study in Brazil. Trans R Soc Trop Med Hyg. 2016;110:199-206. doi: 10.1093/trstmh/trw011.



- **40.** J Carter D, Daniel R, Torrens AW, N Sanchez M, Maciel ELN, Bartholomay P et al. The impact of a cash transfer programme on tuberculosis treatment success rate: a quasi-experimental study in Brazil. BMJ Global Health. 2019;4:e001029. doi: 10.1136/bmjgh-2018-001029.
- **41.** Nery JS, Rodrigues LC, Rasella D, Aquino R, Barreira D, Torrens AW et al. Effect of Brazil's conditional cash transfer programme on tuberculosis incidence. Int J Tuberc Lung Dis. 2017;21:790-6. doi: 10.5588/ijtld.16.0599.
- **42.** Klein K, Bernachea MP, Irribarren S, Gibbons L, Chirico C, Rubinstein F. Evaluation of a social protection policy on tuberculosis treatment outcomes: A prospective cohort study. PLoS Med. 2019;16:e1002788. doi: 10.1371/journal.pmed.1002788.
- **43.** Rocha C, Montoya R, Zevallos K, Curatola A, Ynga W, Franco J et al. The innovative socio-economic interventions against tuberculosis (ISIAT) project: an operational assessment. Int J Tuberc Lung Dis. 2011;15 Suppl 2:50-7. doi: 10.5588/ijtld.10.0447.
- **44.** Wingfield T, Tovar MA, Huff D, Boccia D, Montoya R, Ramos E et al. A randomized controlled study of socioeconomic support to enhance tuberculosis prevention and treatment, Peru. Bull World Health Organ. 2017;95:270-80. doi: 10.2471/BLT.16.170167.
- **45.** Durovni B, Saraceni V, Puppin MS, Tassinari W, Cruz OG, Cavalcante S et al. The impact of the Brazilian Family Health Strategy and the conditional cash transfer on tuberculosis treatment outcomes in Rio de Janeiro: an individual-level analysis of secondary data. J Public Health (Oxf). 2018;40:e359-e66. doi: 10.1093/pubmed/fdx132.
- **46.** Reis-Santos B, Oliosi JGN, Maciel ELN. Effect of the Bolsa Familia Programme on tuberculosis treatment outcomes Authors' reply. Lancet Glob Health. 2019;7:e566. doi: 10.1016/S2214-109X(19)30153-6.
- **47.** Cantalice Filho JP. Food baskets given to tuberculosis patients at a primary health care clinic in the city of Duque de Caxias, Brazil: effect on treatment outcomes. J Bras Pneumol. 2009;35:992-7. doi: 10.1590/s1806-37132009001000008.
- **48.** Garden B, Samarina A, Stavchanskaya I, Alsterlund R, Ovregaard A, Taganova O et al. Food incentives improve adherence to tuberculosis drug treatment among homeless patients in Russia. Scand J Caring Sci. 2013;27:117-22. doi: 10.1111/j. 1471-6712.2012.01009.x.
- **49.** Wingfield T, Boccia D, Tovar MA, Huff D, Montoya R, Lewis JJ et al. Designing and implementing a socioeconomic intervention to enhance TB control: operational evidence from the CRESIPT project in Peru. BMC Public Health. 2015;15:810. doi: 10.1186/s12889-015-2128-0.
- **50.** Sudarsanam TD, John J, Kang G, Mahendri V, Gerrior J, Franciosa M et al. Pilot randomized trial of nutritional supplementation in patients with tuberculosis and HIV-tuberculosis coinfection receiving directly observed short-course chemotherapy for tuberculosis. Trop Med Int Health. 2011;16:699-706. doi: 10.1111/j.1365-3156.2011.02761.x.
- **51.** Lu H, Yan F, Wang W, Wu L, Ma W, Chen J et al. Do transportation subsidies and living allowances improve tuberculosis control outcomes among internal migrants in urban Shanghai, China? Western Pac Surveill Response J. 2013;4:19-24. doi: 10.5365/WPSAR.2013.4.1.003.
- **52.** Zhao Q, Wang L, Tao T, Xu B. Impacts of the "transport subsidy initiative on poor TB patients" in Rural China: a patient-cohort based longitudinal study in rural China. PLoS One. 2013;8:e82503. doi: 10.1371/journal.pone.0082503.
- **53.** Khanal S, Elsey H, King R, Baral SC, Bhatta BR, Newell JN. Development of a patient-centred, psychosocial support intervention for multi-drug-resistant tuberculosis (MDR-TB) care in Nepal. PLoS One. 2017;12:e0167559. doi: 10.1371/journal.pone.0167559.
- **54.** Baral SC, Aryal Y, Bhattrai R, King R, Newell JN. The importance of providing counselling and financial support to patients receiving treatment for multi-drug resistant TB: mixed method qualitative and pilot intervention studies. BMC Public Health. 2014;14:46. doi: 10.1186/1471-2458-14-46.
- **55.** Ghazy RM, El Saeh HM, Abdulaziz S, Hammouda EA, Elzorkany AM, Khidr H et al. A systematic review and meta-analysis of the catastrophic costs incurred by tuberculosis patients. Sci Rep. 2022;12:558. doi: 10.1038/s41598-021-04345-x.
- **56.** Fuady A, Houweling TAJ, Mansyur M, Burhan E, Richardus JH. Catastrophic costs due to tuberculosis worsen treatment outcomes: a prospective cohort study in Indonesia. Trans R Soc Trop Med Hyg. 2020;114:666-73. doi: 10.1093/trstmh/traa038.
- **57.** Mauch V, Bonsu F, Gyapong M, Awini E, Suarez P, Marcelino B et al. Free tuberculosis diagnosis and treatment are not enough: patient cost evidence from three continents. Int J Tuberc Lung Dis. 2013;17:381-7. doi: 10.5588/ijtld.12.0368.
- **58.** Tanimura T, Jaramillo E, Weil D, Raviglione M, Lonnroth K. Financial burden for tuberculosis patients in low- and middleincome countries: a systematic review. Eur Respir J. 2014;43:1763-75. doi: 10.1183/09031936.00193413.
- **59.** Wingfield T, Boccia D, Tovar M, Gavino A, Zevallos K, Montoya R et al. Defining catastrophic costs and comparing their importance for adverse tuberculosis outcome with multi-drug resistance: a prospective cohort study, Peru. PLoS Med. 2014;11:e1001675. doi: 10.1371/journal.pmed.1001675.
- **60.** Wingfield T, Tovar MA, Huff D, Boccia D, Montoya R, Ramos E et al. The economic effects of supporting tuberculosisaffected households in Peru. Eur Respir J. 2016;48:1396-410. doi: 10.1183/13993003.00066-2016.

- **61.** Fuady A, Houweling TAJ, Mansyur M, Burhan E, Richardus JH. Effect of financial support on reducing the incidence of catastrophic costs among tuberculosis-affected households in Indonesia: eight simulated scenarios. Infect Dis Poverty. 2019;8:10. doi: 10.1186/s40249-019-0519-7.
- 62. Rudgard WE, Evans CA, Sweeney S, Wingfield T, Lonnroth K, Barreira D et al. Comparison of two cash transfer strategies to prevent catastrophic costs for poor tuberculosis-affected households in low- and middle-income countries: An economic modelling study. PLoS Med. 2017;14:e1002418. doi: 10.1371/journal.pmed.1002418.
- **63.** WHO consolidated guidelines on tuberculosis. Module 5: management of tuberculosis in children and adolescents. Geneva: World Health Organization; 2022 (https://apps.who.int/iris/handle/10665/352522, accessed 31 August 2023).
- **64.** Atkins S, Heimo L, Carter DJ, Ribas Closa M, Vanleeuw L, Chenciner L et al. The socioeconomic impact of tuberculosis on children and adolescents: a scoping review and conceptual framework. BMC Public Health. 2022;22:2153. doi: 10.1186/s12889-022-14579-7.
- **65.** HIV and social protection assessment tool: generating evidence for policy and action on HIV and social protection. Geneva: UNAIDS; 2017 (https://www.unaids.org/sites/default/files/media_asset/HIV-social-protection-assessment-tool_en.pdf, accessed 31 August 2023).
- **66.** Social protection assessment-based national dialogue: a global guide. Joint United Nations response to implement social protection floors and achieve the Sustainable Development Goals. Geneva: International Labour Organization; 2015 (https://www.social-protection.org/gimi/RessourcePDF.action?id=53462, accessed 31 August 2023).
- **67.** UNDG social protection coordination toolkit: coordinating the design and implementation of nationally defined social protection floors. Geneva: International Labour Organization, United Nations Development Group; 2016 (https://www.ilo.org/wcmsp5/groups/public/---dgreports/---integration/documents/publication/wcms_568699.pdf, accessed 31 August 2023).
- **68.** Inter-Agency Social Protection Assessment tools [website]. Geneva: International Labour Organization; 2023 (https://www.social-protection.org/gimi/gess/ShowWiki.action?wiki.wikiId=2361#tabs-1, accessed 31 August 2023).
- **69.** Guidance for national strategic planning for tuberculosis. Geneva: World Health Organization; 2022 (https://apps.who. int/iris/rest/bitstreams/1458000/retrieve, accessed 31 August 2023).
- **70.** Tuberculosis patient cost surveys: a handbook. Geneva: World Health Organization; 2017 (https://apps.who.int/iris/rest/bitstreams/1092601/retrieve, accessed 31 August 2023).
- **71.** Mont D, Cote A. Inclusive social protection for empowering persons with disabilities: considering the disability related extra costs in social protection. Advanced unedited draft for consultation. Geneva: International Labour Organization, UNICEF, International Disability Alliance, UN Partnership for the Rights of Persons with Disabilities; 2020 (https://www.social-protection.org/gimi/RessourcePDF.action?id=56925, accessed 31 August 2023).
- **72.** ILO social security standards: learn, ratify and apply. Geneva: International Labour Organization; 2020 (https://www. social-protection.org/gimi/RessourcePDF.action?id=55196, accessed 31 August 2023).
- **73.** World Social Protection Data Dashboards: universal social protection to achieve the Sustainable Development Goals [website]. Geneva: International Labour Organization; 2023 (https://www.social-protection.org/gimi/WSPDB. action?id=32, accessed 6 September 2023).
- **74.** Social Protection Monitor: announced measures throughout the world [website]. Geneva: International Labour Organization; 2023 (https://www.social-protection.org/gimi/ShowWiki.action?id=3426, accessed 6 September 2023).
- **75.** Social Protection Response to the COVID-19 Crisis [website]. Geneva: International Labour Organization; 2021 (https://www.social-protection.org/gimi/WSPDB.action?id=62, accessed 6 September 2023).
- **76.** ILO Social Protection Outlook [website]. Geneva: International Labour Organization; 2022 (https://www.social-protection. org/gimi/ShowNewsletter.action, accessed 6 September 2023).
- **77.** Social Security Programs Throughout the World. Country profiles: latest country profile publications [website]. Geneva: International Social Security Association; 2019 (https://ww1.issa.int/ssptw, accessed 6 September 2023).
- **78.** Boccia D, Hargreaves J, Howe LD, De Stavola BL, Fielding K, Ayles H et al. The measurement of household socioeconomic position in tuberculosis prevalence surveys: a sensitivity analysis. Int J Tuberc Lung Dis. 2013;17:39-45. doi: 10.5588/ijtld.11.0387.
- **79.** Framework for collaborative action on tuberculosis and comorbidities. Geneva: World Health Organization; 2022 (https://apps.who.int/iris/rest/bitstreams/1461662/retrieve, accessed 31 August 2023).
- **80.** National surveys of costs faced by tuberculosis patients and their households 2015–2021. Geneva: World Health Organization; 2022 (https://apps.who.int/iris/rest/bitstreams/1492554/retrieve, accessed 31 August 2023).
- **81.** Sekhon M, Cartwright M, Francis JJ. Acceptability of healthcare interventions: an overview of reviews and development of a theoretical framework. BMC Health Serv Res. 2017;17:88. doi: 10.1186/s12913-017-2031-8.



- **82.** South African Government: disability grant [website]. Government of South Africa; 2023 (https://www.gov.za/services/ social-benefits/disability-grant, accessed 6 September 2023).
- **83.** Vanleeuw L, Zembe-Mkabile W, Atkins S. "I'm suffering for food": Food insecurity and access to social protection for TB patients and their households in Cape Town, South Africa. PLoS One. 2022;17:e0266356. doi: 10.1371/journal.pone.0266356.
- **84.** Foster N, Vassall A, Cleary S, Cunnama L, Churchyard G, Sinanovic E. The economic burden of TB diagnosis and treatment in South Africa. Soc Sci Med. 2015;130:42-50. doi: 10.1016/j.socscimed.2015.01.046.
- 85. Bono Joaquín Gallegos Lara beneficia a cerca de 23 mil cuidadores de personas con discapacidad severa [Joaquín Gallegos Lara voucher benefits close to 23 thousand caregivers of people with severe disabilities]. Gobierno de la República del Ecuador [Government of Ecuador]; 2012 (https://www.presidencia.gob.ec/bono-joaquin-gallegos-lara-beneficia-a-cerca-de-23-mil-cuidadores-de-personas-con-discapacidad-severa/, accessed 6 September 2023).
- **86.** Women and men in the informal economy: a statistical picture (third edition). Geneva: International Labour Organization; 2018 (https://www.ilo.org/wcmsp5/groups/public/---dgreports/---dcomm/documents/publication/wcms_626831.pdf, accessed 31 August 2023).
- 87. Extending social health protection: accelerating progress towards universal health coverage in Asia and the Pacific. Geneva: International Labour Organization; 2021 (https://www.ilo.org/wcmsp5/groups/public/---asia/---ro-bangkok/ documents/publication/wcms_831137.pdf, accessed 31 August 2023).
- **88.** Extending social security coverage to workers in the informal economy: lessons from international experience. Geneva: International Labour Organization; 2021 (https://www.ilo.org/wcmsp5/groups/public/---ed_protect/---soc_sec/documents/publication/wcms_749431.pdf, accessed 31 August 2023).
- **89.** Kaswa M, Minga G, Nkiere N, Mingiedi B, Eloko G, Nguhiu P et al. The economic burden of TB-affected households in DR Congo. Int J Tuberc Lung Dis. 2021;25:923-32. doi: 10.5588/ijtld.21.0182.
- **90.** Chittamany P, Yamanaka T, Suthepmany S, Sorsavanh T, Siphanthong P, Sebert J et al. First national tuberculosis patient cost survey in Lao People's Democratic Republic: Assessment of the financial burden faced by TB-affected households and the comparisons by drug-resistance and HIV status. PLoS One. 2020;15:e0241862. doi: 10.1371/journal.pone.0241862.
- **91.** C111 Discrimination (Employment and Occupation) Convention, 1958 (No. 111). Geneva: International Labour Organization; 1958 (https://www.ilo.org/dyn/normlex/en/f?p=NORMLEXPUB:12100:0::NO::P12100_Ilo_Code:C111, accessed 6 September 2023).
- **92.** C158 Termination of Employment Convention, 1982 (No. 158). Geneva: International Labour Organization; 1982 (https://www.ilo.org/dyn/normlex/en/f?p=NORMLEXPUB:12100:0::NO::P12100_ILO_CODE:C158, accessed 6 September 2023).
- **93.** R200 HIV and AIDS Recommendation, 2010 (No. 200). Geneva: International Labour Organization; 2010 (https://www.ilo.org/dyn/normlex/en/f?p=1000:12100:::NO:12100:P12100_INSTRUMENT_ID:2551501, accessed 6 September 2023).
- 94. Bolton R, Whelan J, Dukelow F. What can welfare stigma do? Social Policy and Society. 2022;21:632-45. doi: 10.1017/S1474746422000185.
- **95.** Van der Water N, Yassi A, Ehrlich R. Workers' compensation claims for occupational tuberculosis in South African health workers: outcomes and workers' experiences. S Afr Med J. 2020;110:389-95. doi: 10.7196/SAMJ.2020.v110i5.14247.
- 96. The joint WHO-ILO-UNAIDS policy guidelines on improving health workers' access to HIV and TB prevention, treatment, care and support services: a guidance note. Geneva: International Labour Organization, Joint United Nations Programme on HIV/AIDS and World Health Organization; 2011 (https://www.ilo.org/wcmsp5/groups/public/---ed_protect/---protrav/---ilo_aids/documents/publication/wcms_149714.pdf, accessed 31 August 2023).
- 97. Barreto ML, Ichihara MY, Pescarini JM, Ali MS, Borges GL, Fiaccone RL et al. Cohort profile: the 100 million Brazilian cohort. Int J Epidemiol. 2022;51:e27-e38. doi: 10.1093/ije/dyab213.
- **98.** WHO operational handbook on tuberculosis. Module 4: treatment tuberculosis care and support. Geneva: World Health Organization; 2022 (https://apps.who.int/iris/rest/bitstreams/1446735/retrieve, accessed 31 August 2023).
- **99.** Foster I, Galloway M, Human W, Hanlie MA, Nosivuyile Vanqa M, Wademan DT et al. Analysing interventions designed to reduce tuberculosis-related stigma: A scoping review. PloS Global Health. 2022;2:e0000989. doi: 10.1371/journal.pgph.0000989.
- 100. Rai B, Dixit K, Aryal TP, Mishra G, Siqueira-Filha NT, Paudel PR et al. Developing feasible, locally appropriate socioeconomic support for TB-affected households in Nepal. Trop Med Infect Dis. 2020;5. doi: 10.3390/tropicalmed5020098.
- **101.** Cluver LD, Orkin FM, Boyes ME, Sherr L. Cash plus care: social protection cumulatively mitigates HIV-risk behaviour among adolescents in South Africa. AIDS. 2014;28 Suppl 3:S389-97. doi: 10.1097/QAD.0000000000340.

- **102.** Pettifor A, Wamoyi J, Balvanz P, Gichane MW, Maman S. Cash plus: exploring the mechanisms through which a cash transfer plus financial education programme in Tanzania reduced HIV risk for adolescent girls and young women. J Int AIDS Soc. 2019;22 Suppl 4:e25316. doi: 10.1002/jia2.25316.
- 103. Roelen K, Devereux S, Abdulai A, Martorano B, Palermo T, Ragno LP. How to make 'cash plus' work: linking cash transfers to services and sectors, Innocenti Working Paper no. 2017-10. Florence: UNICEF Office of Research; 2017 (https://www.unicef-irc.org/publications/pdf/IDS%20WP%20Rev%20Jan%202018.pdf, accessed 31 August 2023).
- **104.** Sinha P, Lonnroth K, Bhargava A, Heysell SK, Sarkar S, Salgame P et al. Food for thought: addressing undernutrition to end tuberculosis. Lancet Infect Dis. 2021;21:e318-e25. doi: 10.1016/S1473-3099(20)30792-1.
- **105.** Darnton-Hill I, Mandal PP, de Silva A, Bhatia V, Sharma M. Opportunities to prevent and manage undernutrition to amplify efforts to end TB. Int J Tuberc Lung Dis. 2022;26:6-11. doi: 10.5588/ijtld.21.0488.
- **106.** Lonnroth K, Jaramillo E, Williams BG, Dye C, Raviglione M. Drivers of tuberculosis epidemics: the role of risk factors and social determinants. Soc Sci Med. 2009;68:2240-6. doi: 10.1016/j.socscimed.2009.03.041.
- **107.** Aibana O, Acharya X, Huang CC, Becerra MC, Galea JT, Chiang SS et al. Nutritional status and tuberculosis risk in adult and pediatric household contacts. PLoS One. 2016;11:e0166333. doi: 10.1371/journal.pone.0166333.
- 108. Zachariah R, Spielmann MP, Harries AD, Salaniponi FM. Moderate to severe malnutrition in patients with tuberculosis is a risk factor associated with early death. Trans R Soc Trop Med Hyg. 2002;96:291-4. doi: 10.1016/ s0035-9203(02)90103-3.
- **109.** Bhargava A, Chatterjee M, Jain Y, Chatterjee B, Kataria A, Bhargava M et al. Nutritional status of adult patients with pulmonary tuberculosis in rural central India and its association with mortality. PLoS One. 2013;8:e77979. doi: 10.1371/journal.pone.0077979.
- **110.** Ayiraveetil R, Sarkar S, Chinnakali P, Jeyashree K, Vijayageetha M, Thekkur P et al. Household food insecurity among patients with pulmonary tuberculosis and its associated factors in South India: a cross-sectional analysis. BMJ Open. 2020;10:e033798. doi: 10.1136/bmjopen-2019-033798.
- 111. Hill-Briggs F, Adler NE, Berkowitz SA, Chin MH, Gary-Webb TL, Navas-Acien A et al. Social determinants of health and diabetes: a scientific review. Diabetes Care. 2020;44:258-79. doi: 10.2337/dci20-0053.
- **112.** Myers CA, Beyl RA, Martin CK, Broyles ST, Katzmarzyk PT. Psychological mechanisms associated with food security status and BMI in adults: a mixed methods study. Public Health Nutr. 2020;23:2501-11. doi: 10.1017/S1368980020000889.
- **113.** Rehm J, Gmel GE, Sr., Gmel G, Hasan OSM, Imtiaz S, Popova S et al. The relationship between different dimensions of alcohol use and the burden of disease-an update. Addiction. 2017;112:968-1001. doi: 10.1111/add.13757.
- **114.** Benzekri NA, Sambou J, Diaw B, Sall el HI, Sall F, Niang A et al. High prevalence of severe food insecurity and malnutrition among HIV-infected adults in Senegal, West Africa. PLoS One. 2015;10:e0141819. doi: 10.1371/journal.pone.0141819.
- **115.** Samuel B, Volkmann T, Cornelius S, Mukhopadhay S, MejoJose, Mitra K et al. Relationship between nutritional support and tuberculosis treatment outcomes in West Bengal, India. J Tuberc Res. 2016;4:213-9. doi: 10.4236/jtr.2016.44023.
- **116.** Bock NN, Sales RM, Rogers T, DeVoe B. A spoonful of sugar...: improving adherence to tuberculosis treatment using financial incentives. Int J Tuberc Lung Dis. 2001;5:96-8.
- 117. Mansour O, Masini EO, Kim BJ, Kamene M, Githiomi MM, Hanson CL. Impact of a national nutritional support programme on loss to follow-up after tuberculosis diagnosis in Kenya. Int J Tuberc Lung Dis. 2018;22:649-54. doi: 10.5588/ijtld.17.0537.
- **118.** Guideline: nutritional care and support for patients with tuberculosis. Geneva: World Health Organization; 2013 (https://apps.who.int/iris/rest/bitstreams/443197/retrieve, accessed 31 August 2023).
- **119.** Semba RD, Darnton-Hill I, de Pee S. Addressing tuberculosis in the context of malnutrition and HIV coinfection. Food Nutr Bull. 2010;31:S345-64. doi: 10.1177/15648265100314S404.
- 120. McQuaid CF, Sinha P, Bhargava M, Weerasuriya C, Houben R, Bhargava A. Tuberculosis and nutrition: what gets measured gets managed. Lancet Respir Med. 2023. doi: 10.1016/s2213-2600(23)00009-7.
- 121. Baum F, MacDougall C, Smith D. Participatory action research. J Epidemiol Community Health. 2006;60:854-7. doi: 10.1136/jech.2004.028662.
- 122. Ryckembusch D, Frega R, Silva M, Gentilini U, Grede N, Brown L et al. Enhancing Nutrition: A New Tool for Ex-Ante Comparison of Commodity-based Vouchers and Food Transfers. World Development. 2013;Vol. 49:55-67. doi: 10.1016/j. worlddev.2013.01.021.
- **123.** Martins N, Morris P, Kelly PM. Food incentives to improve completion of tuberculosis treatment: randomised controlled trial in Dili, Timor-Leste. BMJ. 2009;339:b4248. doi: 10.1136/bmj.b4248.
- 124. TB Photovoice [website]. Snohomish, Washington2023 (https://tbphotovoice.org/, accessed 6 September 2023).



- **125.** Ferris France N, Hodgson I, Ramakant B, Macdonald SH, Mitchell EMH. From the inside out: dealing with TB-related self-stigma and shame. A pilot toolkit for people with TB to deal with self-stigma and shame. KNCV; 2018 (https://www.kncvtbc.org/uploaded/2018/10/From-the-Inside-out-stigma-compressed.pdf, accessed 31 August 2023).
- **126.** Leimane L, Mitchell EMH, van de Berg S, Mirtskhulava V. Tuberculosis stigma reduction for health care institutions: the allies approach intervention package. The KNCV stigma reduction tool box. KNCV; 2018 (https://www.kncvtbc.org/uploaded/2018/10/Allies_Approach_V4.pdf, accessed 31 August 2023).
- **127.** Quesada A. The right to breathe: human rights training for TB survivors and affected communities. Manual for facilitators. Bangkok, Thailand: Activists Coalition on Tuberculosis Asia-Pacific; 2019 (https://apcaso.org/wp-content/uploads/2020/02/The-Right-To-Breathe-FINAL.pdf, accessed 31 August 2023).
- **128.** Kidd R, Clay S, Belemu S, Bond V, Chonta M, Chiiya C. Understanding and challenging TB stigma. Toolkit for action. STAMPP, CREATE and International HIV/AIDS Alliance; 2009 (https://assets.publishing.service.gov.uk/media/57a08b3ced915d622c000bb7/TB_and_Stigma_May09.pdf, accessed 31 August 2023).
- **129.** TB stigma assessment tool [website]. Geneva: Stop TB Partnership; 2020 (https://www.stoptb.org/tb-stigma/tb-stigma-assessment-tool, accessed 31 August 2023).
- **130.** Mitchell EMH, van den Hof S. TB stigma measurement guidance. The Hague: KNCV; 2018 (https://www.challengetb.org/publications/tools/ua/TB_Stigma_Measurement_Guidance.pdf, accessed 31 August 2023).
- 131. Yates TA, Ayles H, Leacy FP, Schaap A, Boccia D, Beyers N et al. Socio-economic gradients in prevalent tuberculosis in Zambia and the Western Cape of South Africa. Trop Med Int Health. 2018;23:375-90. doi: 10.1111/tmi.13038.
- **132.** Technical guidance for WFP. consolidated approach for reporting indicators of food security (CARI). Third edition. Rome: World Food Programme; 2021 (https://docs.wfp.org/api/documents/WFP-0000134704/download/?_ga=2.213725969.441296502.1693491698-1187277635.1693491698, accessed 31 August 2023).
- **133.** Herriot M, Valentine NB. Health in all policies as part of the primary health care agenda on multisectoral action. In: Technical series on primary health care. Geneva: World Health Organization; 2018 (https://apps.who.int/iris/ handle/10665/326463, accessed 31 August 2023).
- **134.** Health in all policies: Helsinki statement. Framework for country action. Geneva: World Health Organization; 2014 (https://apps.who.int/iris/rest/bitstreams/515645/retrieve, accessed 31 August 2023).
- **135.** Bell B, Hilson M. Report of a conference on intersectoral action for health: a cornerstone for health-for-all in the twenty-first century. In: Proceedings. Intersectoral Action for Health, Halifax, Canada, 20–23 April 1997: World Health Organization; 1997: (https://apps.who.int/iris/bitstream/handle/10665/63657/WHO_PPE_PAC_97.6.pdf?sequence=1&isAllowed=y, accessed 31 August 2023).
- **136.** Control de la tuberculosis en grandes ciudades de Latinoamérica y el Caribe. Lecciones aprendidas. Washington, DC: Organización Panamericana de la Salud, Organización Mundial de la Salud; 2014 (https://iris.paho.org/handle/10665.2/33988, accessed 30 August 2023).
- **137.** WHO Multisectoral Accountability Framework for TB (MAF-TB) baseline assessment checklist for country use in pursuing a national MAF-TB. Geneva: World Health Organization; 2020 (https://www.who.int/publications/m/item/who-multisectoral-accountability-framework-for-tb-(maf-tb)-baseline-assessment-checklist-for-country-use-in-pursuing-a-national-maf-tb, accessed 31 August 2023).
- **138.** Global tuberculosis report 2021. Geneva: World Health Organization; 2021 (https://apps.who.int/iris/rest/bitstreams/1379788/retrieve, accessed 31 August 2023).
- **139.** Adaptation and implementation of WHO's multisectoral accountability framework to end TB (MAF-TB): best practices. Geneva: World Health Organization; 2022 (https://apps.who.int/iris/rest/bitstreams/1490114/retrieve, accessed 31 August 2023).
- 140. Adaptation and implementation of WHO's multisectoral accountability framework to end TB (MAF-TB): operational guidance. Geneva: World Health Organization; 2023. Licence: CC BY-NC-SA 3.0 IGO.
- 141. Herriot M, Valentine NB. Health in All Policies as part of the primary health care agenda on multisectoral action. In: WHO Discussion Paper Series on Social Determinants of Health, no 11. Geneva: World Health Organization; 2018 (https://apps.who.int/iris/handle/10665/326463, accessed 31 August 2023).
- 142. Adelaide Statement II on Health in All Policies. In: Proceedings. The International Conference on Health in All Policies, Adelaide, Australia: World Health Organization, Government of South Australia; (https://apps.who.int/iris/rest/bitstreams/1273060/retrieve, accessed 31 August 2023).
- 143. Closing the gap in a generation. Health equity through action on the social determinants of health: final report of the commission on social determinants of health. Geneva: World Health Organization; 2008 (https://apps.who.int/iris/ bitstream/handle/10665/43943/9789241563703_eng.pdf, accessed 31 August 2023).
- 144. Social determinants of health [website]. Geneva: World Health Organization; 2023 (https://www.who.int/health-topics/ social-determinants-of-health#tab=tab_1, accessed 6 September 2023).

- **145.** The Southern Africa TB in the Mining Sector Initiative [website]. Washington DC: World Bank; 2023 (https://www.worldbank.org/en/programs/the-southern-africa-tb-in-the-mining-sector-initiative, accessed 1 September 2023).
- 146. Hartel LA, Yazbeck AS, Osewe PL. Responding to health system failure on tuberculosis in Southern Africa. Health Systems & Reform. 2018;4:93-100. doi: 10.1080/23288604.2018.1441621.
- 147. Milman HM, Castillo CA, Sansotta AT, Delpiano PV, Murray J. Scaling up an early childhood development programme through a national multisectoral approach to social protection: lessons from Chile Crece Contigo. BMJ. 2018;363:k4513. doi: 10.1136/bmj.k4513.
- **148.** Peña E, Gonzales S, Minaya E, Tawara N, Linarez O, Gil C. Evaluación del trabajo intersectorial en el manejo de la Tuberculosis en el distrito de SJL / Balance, desafío de ocho años [Evaluation of intersectoral work in the management of tuberculosis in the sjl district / Balance, eight year challenge]. Revista Ciencia y Arte de Enfermería. 2018;3:60-7. doi: 10.24314/rcae.2018.v3n1.12.
- **149.** Solar O, Frenz P. Lessons from Chile's use of system-level theory of change to implement a policy redesign process to address health inequities. New Directions for Evaluation. 2017;2017:101-13. doi: 10.1002/ev.20246.
- 150. Recomendação № 003, de 17 de março de 2011 [Recommendation #003, of 17 March 2011] Ducentésima Décima Nona Reunião Ordinária [Nineteenth Ordinary Meeting]: (https://conselho.saude.gov.br/recomendacoes/2011/Reco003.doc, accessed 20 September 2023).
- **151.** Rincón-Torres CE, Rubio V, Castro C, García I, Cruz OA, Trujillo-Trujillo J et al. Red Nacional de Gestión de Conocimiento, Investigación e Innovación en tuberculosis en Colombia [National Network for Knowledge Management, Research and Innovation in tuberculosis in Colombia]. Rev Panam Salud Publica. 2021;45:e23. doi: 10.26633/rpsp.2021.23.
- **152.** Wales and the Sustainable Development Goals 2019. Supplementary report to the United Kingdom of Great Britain and Northern Ireland Voluntary National Review of progress towards the Sustainable Development Goals 2030. Wales, U.K.: Llywodraeth Cymru [Government of Wales]; 2019 (https://www.gov.wales/sites/default/files/publications/2019-07/ supplementary-report-to-the-uk-review-of-progress-towards-the-sustainable-development-goals-2030_0.pdf, accessed 20 September 2023).
- **153.** Progressing the Sustainable Development Goals through Health in All Policies: case studies from around the world. Adelaide, Australia: Government of South Australia, World Health Organization; 2017 (https://cdn.who.int/media/docs/ default-source/inaugural-who-partners-forum/progressing-sdg-case-studies-20178560b677-aeef-4dea-a6cc-459258712d41.pdf?sfvrsn=d0096b17_1&download=true, accessed 1 September 2023).
- **154.** Hendry A, Thompson M, Knight P, McCallum E, Taylor A, Rainey H et al. Health and social care reform in Scotland what next? Int J Integr Care. 2021;21:7. doi: 10.5334/ijic.5633.
- **155.** Cramm JM, Finkenflügel HJ, Møller V, Nieboer AP. TB treatment initiation and adherence in a South African community influenced more by perceptions than by knowledge of tuberculosis. BMC Public Health. 2010;10:72. doi: 10.1186/1471-2458-10-72.
- **156.** Financing across sectors for sustainable development. Guidance note. New York: United Nations Development Programme; 2019 (https://www.undp.org/sites/g/files/zskgke326/files/publications/Guidance%20Note_Financing%20 across%20sectors%20for%20sustainable%20development.pdf, accessed 1 September 2023).
- **157.** Jaramillo CAA, Granados SAA, Alvarez Higuita S. Design of the intersectoral and interinstitutional strategy for the prevention of TB. Administración & Desarrollo. 2020;50:53-70.

Annex 1. Summary of healthrelated social protection policies, guidelines and guidance developed by the UN

This annex provides a comprehensive, but non-systematic overview of some of the most prominent policies, guidelines and guidance on social protection and health developed by UN agencies that collaborate closely with WHO, namely the ILO, WFP, FAO, UNICEF, the United Nations High Commissioner for Refugees (UNHCR), and the IOM.

The annex aims to add to the information provided in this guidance by listing a selection of essential reading for NTP staff and healthcare workers willing to strengthen their knowledge on the impact and implementation of social protection programmes to address major public health challenges. While many of the references listed here do not necessarily pertain to TB, indirect lessons can be still transferred to TB.

Key references have been grouped under the following thematic areas:

- 1. Social health protection in the workplace
- 2. Social protection and nutrition
- **3.** Social protection for key health outcomes including TB, HIV/AIDS, malaria, mental health, child and maternal health and COVID-19
- 4. Social protection and migration health

1. Social health protection and health promotion in the workplace

- Health and Wellness at Work: guidelines for an integrated health testing approach under VCT@WORK. Geneva: International Labour Organization; 2020. https://www.ilo.org/wcmsp5/groups/public/---dgreports/---gender/documents/publication/ wcms_762676.pdf
- Sickness benefits during sick leave and quarantine: country responses and policy considerations in the context of COVID-19. Geneva: International Labour Organization; 2020. https://www.ilo.org/wcmsp5/groups/public/---ed_protect/---soc_sec/documents/publication/wcms_744510.pdf
- Social health protection [website]. Geneva: International Labour Organization; 2023. https://www.ilo.org/secsoc/areas-of-work/policy-development-and-applied-research/social-healthprotection/lang--en/index.htm

- Working together with businesses: guidance on TB and TB/HIV prevention, diagnosis, treatment and care in the workplace. Geneva: World Health Organization, Joint United Nations Programme on HIV/ AIDS (UNAIDS), International Labour Organization; 2012. https://apps.who.int/iris/handle/10665/44833
- Joint WHO/ILO policy guidelines on improving health worker access to prevention, treatment and care services for HIV and TB. Geneva: World Health Organization, International Labour Organization; 2010. https://apps.who.int/iris/handle/10665/44467 https://www.ilo.org/sector/Resources/codes-of-practice-and-guidelines/WCMS_160898/lang--en/index. htm
- The joint WHO-ILO-UNAIDS policy quidelines on improving health workers' access to HIV and TB prevention, treatment, care and support services: a guidance note. Geneva: International Labour Organization; 2011. https://www.ilo.org/global/topics/hiv-aids/publications/WCMS_149714/lang--en/index.htm
- Maher D, Boldrini F, Pathania V, Alli B, Gabriel P, et al. Guidelines for workplace TB control activities: the contribution of workplace TB control activities to TB control in the community. Geneva: World Health Organization, International Labour Organization; 2003. https://apps.who.int/iris/handle/10665/42704
- Recommendation concerning HIV and AIDS and the World of Work, 2010 (No. 200). International Labour Conference 99th Session. Geneva: International Labour Organization; 2010. https://www.ilo.org/global/topics/hiv-aids/WCMS_142706/lang--en/index.htm

2. Social protection and nutrition

- Building the blocks for nutrition-sensitive social protection systems in Asia: informing design, prompting *implementation*. Bangkok: World Food Programme; 2017. https://docs.wfp.org/api/documents/WFP-0000022602/download/
- World Food Programme strategy for support to social protection. Rome: World Food Programme; 2021. https://www.wfp.org/publications/world-food-programme-strategy-support-social-protection-2021
- Nutrition and Social Protection. Rome: Food and Agriculture Organization of the United Nations; 2015. www.fao.org/3/a-i4819e.pdf
- Strengthening sector policies for better food security and nutrition results: social protection. Policy quidance note 4. Rome: Food and Agriculture Organization of the United Nations, Directorate for International Cooperation and Development of the European Commission to boost food and nutrition security, sustainable agriculture and resilience: 2017.

https://www.fao.org/policy-support/tools-and-publications/resources-details/en/c/885771/
Annex 1



- Update on WFP's response to HIV and AIDS. In: Proceedings. World Food Programme Executive Board Annual session, Rome, 10–14 June 2019; WFP/EB.A/2019/5-F. https://executiveboard.wfp.org/document_download/WFP-0000104659
- Leaving no-one behind how WFP's approach to HIV-sensitive social protection will help to achieve zero hunger in Easter and Southern Africa. Policy Brief. Rome: World Food Programme and the UKRI GCRF Accelerating Achievement for Africa's Adolescents (Accelerate) Hub; 2019. https://docs.wfp.org/api/documents/WFP-0000108479/download/?_ ga=2.163479033.411467165.1668700711-1700313855.1668700711
- Providing cash transfers for vulnerable people living with HIV and key populations: lessons learned from a World Food Programme and UNAIDS initiative to mitigate the impact of COVID-19 in western and central Africa. Geneva: Joint United Nations Programme on HIV/AIDS (UNAIDS), World Food Programme; 2022. https://docs.wfp.org/api/documents/WFP-0000144277/download/?_ ga=2.57700975.411467165.1668700711-1700313855.1668700711
- How to make social protection inclusive of people living with HIV and key populations: a checklist. Geneva: United Nations Development Programme, International Labour Organization; 2022. https://www.ilo.org/wcmsp5/groups/public/---dgreports/---gender/documents/publication/ wcms_852400.pdf
- Global Dialogue and Online Consultation on Social Protection for People Living with HIV and Key Populations Most at Risk of HIV [website]. Geneva: United Nations Development Programme, International Labour Organization; 2021. https://www.sparkblue.org/PLHIV2021
- HIV and social protection guidance note. Geneva: Joint United Nations Programme on HIV/ AIDS (UNAIDS); 2014. https://www.unaids.org/sites/default/files/media_asset/2014unaidsguidancenote_HIVandsocialprotection_ en.pdf
- Making universal social protection a reality for people living with, at the risk of, and affected by HIV or tuberculosis. Geneva: International Labour Organization; 2021. https://www.ilo.org/wcmsp5/groups/public/---dgreports/---gender/documents/publication/ wcms_830750.pdf
- The role of social protection in the elimination of child labour: evidence review and policy implications. Geneva and Florence: International Labour Organization and UNICEF Office of Research – Innocenti; 2022. https://www.ilo.org/wcmsp5/groups/public/---ed_norm/---ipec/documents/publication/wcms_845168.pdf
- Being prepared and acting first: a series of case studies on UNICEF's role in the delivery of effective social protection responses to COVID-19. New York: UNICEF; 2021. https://www.unicef.org/documents/being-prepared-and-acting-first

- Towards universal social protection for children: achieving SDG 1.3. ILO-UNICEF Joint Report on Social Protection for Children. Geneva and New York: International Labour Organization, UNICEF; 2019. https://www.unicef.org/reports/towards-universal-social-protection-children-achieving-sdg-13
- UNICEF's Global social protection programme framework. A framework for child sensitive social protection. New York: UNICEF; 2019. https://www.unicef.org/reports/global-social-protection-programme-framework-2019

4. Social protection and migration health

- Securing social protection for migrant workers and their families: challenges and options for building a better future. In: Proceedings. International Labour Organization Governing Body 344th Session, Geneva, March 2022; GB.344/POL/1. https://www.ilo.org/gb/GBSessions/GB344/pol/WCMS_837401/lang--en/index.htm
- Handbook on social health protection for refugees: approaches, lessons learned and practical tools to assess coverage options. Geneva: International Labour Organization, United Nations High Commission for Refugees; 2020. https://www.ilo.org/global/programmes-and-projects/prospects/themes/social-protection/ WCMS_760307/lang--en/index.htm
- UNHCR Mapping of Social Safety Nets for Refugees Opportunities and Challenges [website]. Geneva: United Nations High Commission for Refugees; 2018. https://www.unhcr.org/5ad5b4084.pdf

Annex 2. Glossary

Cash transfer programme: non-contributory scheme or programme providing cash benefits to individuals or households, usually financed out of taxation, other government revenue, or external grants or loans. Cash transfer programmes² may or may not include a means test.

Cash transfer programmes that provide cash to families subject to the condition that they fulfil specific behavioural requirements are referred to as **conditional cash transfer programmes (CCTs)**. This may mean, for example, that beneficiaries must ensure their children attend school regularly, or that they utilize basic preventative nutrition and healthcare services.

Contributory social protection programmes: programmes in which contributions made by protected persons (actual or potential beneficiaries) directly determine entitlement to benefits (acquired rights). The most common example of contributory social security programmes are statutory social insurance programmes, usually covering workers in (formal) wage employment and, in some countries, the self-employed, and in many cases also their families and categories of the population exempt from contributions for certain reasons (for example, caring for children).

Coping strategies: strategies or mechanisms which relieve the impact of the risk once it has occurred. The main forms of coping strategies consist of individual dis-saving/borrowing, migration, selling labour (including that of children), selling assets, reduction of food intake, or the reliance on public or private transfers. Governments have an important role in assisting people in coping, for example, where individual households have not saved enough to handle repeated or catastrophic risks, having been poor all their life with no possibility to accumulate assets. Governments should also establish other types of risk management mechanisms as a more efficient way to deal with risks (1).

Covariant risks: risks, or combination of risks, that affect a large number of people at the same time (for example, an earthquake or a major flood) (1).

Employed persons: all persons above a specified age who, during a specified reference period, were in the following categories: (i) paid employment, (ii) at work; or (iii) with a job but temporarily not at work (1).

Employment guarantee scheme: public employment programme which provides a guaranteed number of workdays per year to poor households, generally providing wages at a relatively low level (typically at the minimum wage level if this is adequately defined).

Informal economy: encompasses both work in the informal sector as well as informal work in the formal sector.

The term "**informal sector**" designates units engaged in the production of goods or services with the primary objective of generating employment and incomes to the persons concerned. Such units are unincorporated enterprises not constituted as separate entities independently of their owners (2). Typically, they operate at a low level of organization, on a small scale and with little or no division of labour and capital as factors of production. Therefore, many micro and small enterprises (MSEs), including those operating in the rural economy, are in the informal sector. In some countries, such MSEs are not or are only partially included in the scope of labour and social protection legislation. Similarly, they often face practical difficulties complying with tax, labour and social security legislations due to their limited administrative and financial capacities. In many cases, neither the employers nor the workers in MSEs are covered by social protection systems.

² Strictly speaking, this term would encompass all social transfers provided in cash, including fully or partially contributory transfers, yet it is usually understood as limited to non-contributory transfers.

The term "**informal employment**" is defined in terms of the employment relationship and protections associated with the job of the worker (3). It refers to working arrangements that are de facto or de jure not subject to national labour legislation, income taxation or entitlement to social protection or certain other employment benefits (such as advance notice of dismissal, severance pay, or paid annual or sick leave). Workers in informal employment are defined as those who work in informal jobs, whether carried out in formal sector enterprises, informal sector enterprises, or households. It can encompass diverse realities when it comes to status in employment (such as employees holding informal jobs, employers and own-account workers employed in their own informal sector enterprises, members of informal producers' cooperatives, contributing family workers in formal or informal sector enterprises and own-account workers engaged in the production of goods for own end use by their household (4). Although not everyone in the informal economy is poor, a significant proportion of the poor are in the informal economy (2).

Multidrug-resistant tuberculosis (MDR-TB): TB that does not respond to at least isoniazid and rifampicin, the two most powerful anti-TB drugs.

National Tuberculosis Programme (NTP): the national entity in charge of the response against TB in the country. Typically, the NTP is part of the local ministry of health and provides technical and managerial leadership to achieve the global targets set by the WHO *End TB Strategy (5)*.

Non-contributory social protection programmes: programmes that require no direct contribution from beneficiaries or their employers as a condition of entitlement to receive relevant benefits. The term covers a broad range of programmes, for example:

- universal non-contributory services for all residents (such as national health services);
- **categorical** programmes for certain broad groups of the population (such as universal pensions for all older people above a certain age, or universal child benefits for all children below a certain age);
- means-tested programmes in which only persons below a certain income level are eligible for the benefit (such as family benefits targeted only at poor families); and
- benefit-tested programmes in which only persons who are not already receiving benefits from another
 programme are eligible (for example social pensions for older persons not already receiving a pension from
 a social insurance programme because they were not able to meet the minimum contributory requirements
 over their working life).

Percentage of TB patients and their households facing catastrophic total costs due to TB: an indicator that examines total costs borne by patients receiving TB treatment that exceed 20% of the household's annual pre-TB disease income or expenditure. The measure seeks to determine whether the sum of direct medical expenditures, direct nonmedical expenditures and indirect costs (for example, income losses) exceeds a certain threshold (> 20%) compared with the household's economic resources available to pay for basic subsistence needs (*6*).

Public employment programme: a government programme offering employment opportunities to certain categories of persons who are unable to find other employment. Public employment programmes include employment guarantee schemes and "cash for work" and "food for work" programmes.

Social assistance: a programme that provides benefits to vulnerable groups of the population, especially households living in poverty. Most social assistance programmes are means-tested. Some are targeted based on categories of vulnerability, while others are targeted to low-income households. They can take several forms, for example non-contributory social pensions, fee waivers and exemptions for health care, schooling or utilities. Conditional cash transfers, for example, are programmes that provide cash to families if they fulfil specific behavioural requirements. Beneficiaries may be required to ensure that their children attend school regularly, or



use basic preventive nutrition and healthcare services. Almost all countries at all income levels have at least one form of social assistance programme.

Social determinants of health (SDH): the social determinants of health (SDH) are the non-medical factors that influence health outcomes. They are the conditions in which people are born, grow, work, live, and age, and the wider set of forces and systems shaping the conditions of daily life. These forces and systems include economic policies and systems, development agendas, social norms, social policies and political systems. The following list provides examples of the social determinants of health, which can influence health equity in positive and negative ways:

- Income and social protection
- Education
- Unemployment and job insecurity
- Working life conditions
- Food insecurity
- Housing, basic amenities and the environment
- Early childhood development
- Social inclusion and non-discrimination
- Structural conflict
- Access to affordable health services of decent quality.

Social health protection: a series of public or publicly organized and mandated private measures against social distress and economic loss caused by the reduction of productivity, stoppage or reduction of earnings, or the cost of necessary treatment that can result from ill health *(7)*.

Social insurance: contributory social protection programme that guarantees protection through an insurance mechanism, based on: (1) the payment of contributions before the occurrence of the contingency; (2) the sharing or "pooling" of risk; and (3) the notion of a guarantee. The contributions paid by (or for) insured people are pooled together, and the resulting fund is used to cover the expenses incurred exclusively by those individuals affected by the occurrence of the contingency. In contrast to commercial insurance, risk-pooling in social insurance is based on the principle of solidarity, with contributions typically related to people's capacity to pay (that is, proportional to earnings) as opposed to premiums that reflect individual risks. For example, social health insurance is typically a health insurance programmes with public stewardship and at least some insurance contributions from and/or on the behalf of the protected persons or some categories of protected persons.

Social protection: also referred to as social security, social protection is a human right and is defined as the set of policies and programmes designed to reduce and prevent poverty, vulnerability and social exclusion throughout the life cycle. Social protection includes nine main areas: child and family benefits, maternity protection, unemployment support, employment injury benefits, sickness benefits, health protection (medical care), old-age benefits, invalidity/ disability benefits, and survivors' benefits. Social protection systems address all these policy areas via a mix of contributory schemes (social insurance) and non-contributory tax-financed benefits (including social assistance). "Social protection" is also currently used to refer to the human right to "social security" and the two terms are often used interchangeably. It must be noted that sometimes the term "social protection" is used with a wider variety of meanings than "social security", including protection provided between members of the family or members of a local community; on other occasions it is also used with a narrower meaning, understood as comprising only measures addressed to the poorest, most vulnerable or excluded members of society. The ILO and in general UN institutions use both terms in discourse with their constituents and in the provision of relevant advice to them.

Social protection floor (SPF): ILO Recommendation No. 202 *(8)* sets out that member States should establish and maintain national social protection floors as a nationally defined set of basic social security guarantees which secure protection aimed at preventing or alleviating poverty, vulnerability and social exclusion *(8)*. These guarantees should ensure at a minimum that over the life cycle, all in need have access to at least essential health care and basic income security. These together ensure effective access to essential goods and services defined as necessary at the national level. More specifically, national social protection floors should comprise at least the following four social security guarantees, as defined at the national level:

- A. access to essential health care, including maternity care;
- B. basic income security for children;
- **C.** basic income security for persons in active age who are unable to earn sufficient income, in particular in cases of sickness, unemployment, maternity and disability; and
- D. basic income security for older persons (8).

Such guarantees should be provided to all residents and all children, as defined in national laws and regulations, and subject to existing international obligations.

Recommendation No. 202 also states that basic social security guarantees should be established by law (8). National laws and regulations should specify the range, qualifying conditions and levels of the benefits giving effect to these guarantees and provide for effective and accessible complaint and appeal procedures. Social protection floors correspond in many ways to the existing notion of "core obligations", to ensure the realization of, at the very least, minimum essential levels of rights embodied in human rights treaties (8,9). Recommendation No. 202 (8) sets out that higher levels of protection should be progressively achieved by national social security systems in line with Convention No. 102 (10) and other ILO instruments.

Social protection programme: a distinct framework of rules to provide social protection benefits to entitled beneficiaries. Such rules would specify the geographical and personal scope of the programme (target group), entitlement conditions, the type of benefits, benefit amounts (cash transfers), periodicity and other benefit characteristics, as well as the financing (contributions, general taxation, other sources), governance and administration of the programme. Social protection programmes are often referred to as "social protection schemes". A social protection programme/scheme can be supported by one or more social security institutions governing the provision of benefits and their financing.

Social protection system: Totality of social protection schemes and programmes in a country. All the social protection schemes and institutions in a country are inevitably interlinked and complementary in their objectives, functions and financing, and thus form a national system. For reasons of effectiveness and efficiency, it is essential that there is close coordination within the system, and that financing of the schemes comprising the system is planned in an integrated way.

Social security: The fundamental right to social security is set out in the Universal Declaration on Human Rights (1948) *(11)*, the International Covenant on Economic, Social and Cultural Rights (1966) *(12)*, and in other major UN human rights instruments. States have the legal obligation to protect and promote human rights, including the right to social protection, or social security, and to ensure that people can realize their rights without discrimination. The overall responsibility of the State includes ensuring the due provision of benefits according to clear and transparent eligibility criteria and entitlements, and the proper administration of the institutions and services. Where benefits and services are not provided directly by public institutions, the effective enforcement of the legislative frameworks is particularly important for the provision of benefits and services *(13)*. The notion of social security covers all measures providing benefits, whether in cash or in kind, to secure protection, inter alia, from:



- lack of work-related income (or insufficient income) caused by sickness, disability, maternity, employment injury, unemployment, old age, or death of a family member;
- · lack of (affordable) access to health care;
- insufficient family support, particularly for children and adult dependants; and
- general poverty and social exclusion.

Social security thus has two main (functional) dimensions, namely "income security" and "availability of medical care".³ Access to social security is essentially a public responsibility, and is typically provided through public institutions, financed from either contributions or taxes or both. However, the delivery of social security can be and often is mandated to private entities. Two main features distinguish social security from other social arrangements. First, benefits are provided to beneficiaries without any simultaneous reciprocal obligation (thus it does not, for example, represent remuneration for work or other services delivered). Second, it is not based on an individual agreement between the protected person and the provider (as is, for example, a life insurance contract); the agreement applies to a wider group of people and so has a collective character.

Social transfer. All social protection benefits comprise transfers either in cash or in kind, that is, they represent a transfer of income, goods or services (for example, healthcare services). This transfer may be from the active to the old, the healthy to the sick, or the affluent to the poor, among others. The recipients of such transfers may be in a position to receive them from a specific scheme because they have contributed to such a scheme (contributory scheme), or because they are residents (universal schemes for all residents), or because they fulfil specific age criteria (categorical schemes), or specific resource conditions (social assistance schemes), or because they fulfil several of these conditions at the same time. In addition, it is a requirement in some schemes (employment guarantee schemes, public employment programmes) that beneficiaries accomplish specific tasks or adopt specific behaviours (conditional cash transfer programmes). In any given country, several schemes of different types generally coexist and may provide benefits for similar contingencies to different population groups.

Tuberculosis (TB): TB is caused by bacteria (*Mycobacterium tuberculosis*), and it most often affects the lungs. TB is spread through the air when people with lung TB cough, sneeze or spit. A person needs to inhale only a few germs to become infected. Every year, 10 million people fall ill with TB. Despite being a preventable and curable disease, 1.5 million people die from TB each year – making it the world's top infectious killer. TB is the leading cause of death of people with HIV and is also a major contributor to antimicrobial resistance. Most of the people who fall ill with TB live in low- and middle-income countries, but TB is present all over the world. About half of all people with TB can be found in eight countries: Bangladesh, China, India, Indonesia, Nigeria, Pakistan, the Philippines and South Africa. About a quarter of the global population is estimated to have been infected with TB bacteria, but most people will not go on to develop TB disease and some will clear the infection. Those who are infected but not (yet) ill with the disease cannot transmit it. People infected with TB bacteria have a 5–10% lifetime risk of falling ill with TB. Those with compromised immune systems, such as people living with HIV, malnutrition or diabetes, or people who use tobacco, have a higher risk of falling ill.

TB-specific social protection: programmes focused on people with TB and their households with the specific aim to improve TB prevention, care and support.

³ These two main dimensions are also identified in the ILO Income Security Recommendation, 1944 (No. 67), and the Medical Care Recommendation, 1944 (No. 69), respectively, as "essential element[s] of social security". These Recommendations envisage that, first, "income security schemes should relieve want and prevent destitution by restoring, up to a reasonable level, income which is lost by reason of inability to work (including old age) or to obtain remunerative work or by reason of the death of a breadwinner" (Recommendation No. 67, Guiding principles, Para. 1); and second, that "a medical care service should meet the need of the individual for care by members of the medical and allied professions" and "the medical care service should cover all members of the community" (Recommendation No. 69, Paras 1 and 8). Recommendation No. 202 also reflects these two elements in the basic social protection guarantees that should form part of national social protection floors.

TB-sensitive social protection: programmes that are not limited to household and individuals affected by TB but that can potentially affect the TB epidemic by covering people with TB, at risk of TB, and/or affected by the consequences of TB.

Universal health coverage: access for all to necessary health services (including promotion, prevention, treatment, rehabilitation and palliation) without financial hardship.

WHO *End TB Strategy*: the vision articulated in the *End TB Strategy (5)* is "a world free of TB", also expressed as "zero deaths, disease and suffering due to TB". All countries could use this vision in national strategies and plans. The WHO *End TB Strategy* has set the following global targets:

- 80% reduction in TB incidence by 2030, compared with 2015–2020 milestone, 20% reduction;
- 90% reduction in the number of TB deaths by 2030, compared with 2015–2020 milestone, 35% reduction; and
- no TB-affected households face catastrophic costs by 2020.

Key elements contributing towards reaching these targets include optimum use of existing interventions, achievement of universal health coverage for essential prevention, treatment and care interventions as well as efforts to address the social determinants and consequences of TB and consequences of TB, including TB-associated impairments and disabilities. The three high-level indicators of the *End TB Strategy* – reductions in TB deaths, reductions in the TB incidence rate and elimination of catastrophic costs – are relevant to all countries. However, targets and milestones for these indicators can be adapted by countries to reflect such factors as different starting points, the main drivers of local epidemics, national policy and strategy related to universal health coverage and social protection and planned programmes. Countries need to set their own national targets guided by the global level of ambition but taking into account national circumstances (14).

Annex 2 references

- Social protection assessment-based national dialogue: a global guide. Joint United Nations response to implement social protection floors and achieve the Sustainable Development Goals. Geneva: International Labour Office; 2015 (https://www.social-protection.org/gimi/RessourcePDF.action?id=53462, accessed 31 August 2023).
- 2. Resolution concerning statistics of employment in the informal sector, adopted by the Fifteenth International Conference of Labour Statisticians. In: Proceedings. Fifteenth International Conference of Labour Statisticians, Geneva: International Labour Organization; (http://ilo.org/wcmsp5/groups/public/---dgreports/---stat/documents/normativeinstrument/wcms_087484.pdf, accessed 7 September 2023).

Annex 2

- 3. Guidelines concerning a statistical definition of informal employment. Geneva: International Labour Organization; 2003 (https://ilo.org/wcmsp5/groups/public/---dgreports/---stat/documents/normativeinstrument/wcms_087622.pdf, accessed 7 September 2023).
- **4.** General Report. In: Proceedings. Seventeenth International Conference of Labour Statisticians, Geneva: International Labour Organization; (https://ilo.primo.exlibrisgroup.com/discovery/fulldisplay/alma993953873402676/411LO_INST:411LO_V2, accessed 7 September 2023).
- 5. The end TB strategy. Geneva; 2015 (WHO/HTM/TB/2015.19; https://www.who.int/publications/i/item/WHO-HTM-TB-2015.19, accessed 31 August 2023).
- 6. National surveys of costs faced by tuberculosis patients and their households 2015–2021. Geneva: World Health Organization; 2022 (https://apps.who.int/iris/rest/bitstreams/1492554/retrieve, accessed 31 August 2023).
- **7.** Towards Universal Health Coverage: Social Health Protection Principles. Geneva: International Labour Organization; 2020 (https://www.ilo.org/wcmsp5/groups/public/---ed_protect/---soc_sec/documents/publication/wcms_740724. pdf, accessed 7 September 2023).
- R202 Social Protection Floors Recommendation, 2012 (No. 202). Recommendation concerning National Floors of Social Protection. Geneva: International Labour Organization; 2012 (https://www.ilo.org/dyn/normlex/en/f?p=NORMLEX PUB:12100:0::NO::P12100_INSTRUMENT_ID:3065524, accessed 5 September 2023).
- **9.** Report on social protection floors and economic and social rights. Geneva: Office of the United Nations High Commissioner for Human Rights; 2015 (https://www.ohchr.org/sites/default/files/ReportSocialProtectionFloors.pdf, accessed 21 September 2023).
- C102 Social Security (Minimum Standards) Convention, 1952 (No. 102). Convention concerning minimum standards of social security. Geneva: International Labour Organization; 1952 (https://www.ilo.org/dyn/normlex/en/f?p=NORMLEXPU B:12100:0::NO::P12100_INSTRUMENT_ID:312247, accessed 5 September 2023).
- **11.** The Universal Declaration of Human Rights (Resolution 217A (III)), 10 December 1948. New York: United Nations General Assembly; 1948 (http://www.un.org/en/universal-declaration-human-rights/, accessed 5 September, 2023).
- 12. International Covenant on Economic, Social and Cultural Rights. New York: United Nations General Assembly; 1966 (https://www.ohchr.org/en/instruments-mechanisms/instruments/international-covenant-economic-social-and-cultural-rights, accessed 7 September 2023).
- **13.** General comment no. 19, The right to social security (art. 9). Geneva: United Nations Committee on Economic, Social and Cultural Rights; 2008 (https://digitallibrary.un.org/record/618890, accessed 21 September 2023).
- 14. Implementing the end TB strategy: the essentials, 2022 update. Geneva: World Health Organization; 2022 (https://apps. who.int/iris/rest/bitstreams/1486092/retrieve, accessed 30 August 2023).

Guidance on social protection for people affected by tuberculosis



For further information, please contact:

Global TB Programme World Health Organization 20, Avenue Appia CH-1211 Geneva 27 Switzerland Web site: www.who.int/tb

Universal Social Protection Department International Labour Organization Route des Morillons 4 – CH1211 Geneva 22 Switzerland Website: www.social-protection.org

